ТО:	CHAIR AND MEMBERS COMMUNITY AND PROTECTIVE SERVICES COMMITTEE MEETING ON FEBRUARY 19, 2020
FROM:	SANDRA DATARS BERE MANAGING DIRECTOR, HOUSING, SOCIAL SERVICES AND DEARNESS HOME
SUBJECT:	ONTARIO TRANSFER PAYMENT AGREEMENT SKILLS ADVANCE ONTARIO (SAO) - EMPLOYMENT SERVICES FOR THE MANUFACTURING SECTOR IN LONDON-MIDDLESEX

RECOMMENDATION

That, on the recommendation of the Managing Director, Housing, Social Services and Dearness Home, the <u>attached</u> proposed By-law (Appendix "A") **BE INTRODUCED** at the Municipal Council meeting of March 24, 2020 to:

- 1. AUTHORIZE AND APPROVE the Ontario Transfer Payment Agreement Skills Advance Ontario (SAO) Employment Services for the Manufacturing Sector in London–Middlesex between Her Majesty the Queen in Right of Ontario as represented by the Minister of Labour, Training and Skills Development and The Corporation of the City of London, substantially in the form <a href="https://doi.org/10.1007/jtm2.2007/j
- a) **AUTHORIZE** the Mayor and City Clerk to execute the Ontario Transfer Payment Agreement Skills Advance Ontario (SAO) Employment Services for the Manufacturing Sector in London-Middlesex attached as Schedule "1" to the proposed by-law, for the delivery of Employment Services for the Manufacturing Sector in London-Middlesex; authorized and approved in subsection 1, above.
- b) DELEGATE to the Managing Director, Housing, Social Services and Dearness Home, the
 authority to approve any further Amendments to the Ontario Transfer Payment Agreement
 Skills Advance Ontario (SAO) Employment Services for the Manufacturing Sector in
 London-Middlesex if the Amendments are substantially in the form of the Agreement,
 attached as Schedule "1" to the proposed by-law;
- AUTHORIZE the Managing Director, Housing, Social Services and Dearness Home, or written designate to execute any Amendments to the Ontario Transfer Payment Agreement

 Skills Advance Ontario (SAO) Employment Services for the Manufacturing Sector in London-Middlesex as approved by the proposed by-law; and,
- d) **DELEGATE** to Managing Director, Housing, Social Services and Dearness Home, or written designate, the authority to undertake all the administrative, financial and reporting acts, including signing authority regarding: application forms for funding, budgets, cash flows, other financial reporting including financial claims, and directions, consents and other authorizations as may be required, provided that the monetary amounts do not exceed the maximum amount of the Ministry of Labour, Training and Skills Development's contribution specified in the Agreement that are necessary in connection with the Ontario Transfer Payment Agreement Skills Advance Ontario (SAO) Employment Services for the Manufacturing Sector in London-Middlesex as approved by the proposed by-law.

PREVIOUS REPORTS PERTINENT TO THIS MATTER

Employment Ontario Transformation – Service System Manager Update (CPSC – January 21, 2020)

Employment Ontario Transformation – Service System Manager Competition (CPSC-June 17, 2019)

Purchase of Service Agreements – Ontario Works Employment Assistance Services (CPSC, December 10, 2018)

PURPOSE

The purpose of this report is to recommend approval to enter into a Funding Agreement with the Provincial Government of Ontario under the Skills Advance Ontario pilot project for the period of March 24, 2020 to March 31, 2021.

STRATEGIC PLAN LINKAGES 2019-2023

Strengthening Our Community

- Londoners have access to the supports they need to be successful.
- Londoners have access to the services and supports that promote well-being, health, and safety in their neighbourhoods and across the city.

Growing Our Economy

 London creates a supportive environment where entrepreneurs, businesses, and talent can thrive.

Leading in Public Service

• The City of London is a leader in public service as an employer, a steward of public funds, and an innovator of service.

BACKGROUND

Currently the London Census Metropolitan Area (CMA) has one of the lowest labour market participation rates in Ontario while at the same time job opportunities continue to increase, and local employers in the region experience significant difficulty in filling vacant positions. Upwards of 6,500 vacant positions exist, and this number is expected to rise over the next several years. (This estimate is based on jobs aggregated through the Local Job Hub website with an additional 60% as an estimate of un-posted jobs).

City of London Social Services administers the Ontario Works program on behalf of the Province of Ontario and the Ministry of Children, Community, and Social Services as a Consolidated Municipal Service Manager (CMSM). Ontario Works is an employment assistance and financial support program focused on helping individuals and families gain and maintain sustainable employment through training, education, developing "hard" and "soft" employment skills, linking participants to opportunities, and providing individualized supports as needed. The City of London is responsible for the effective delivery of individualized service and supports that respond to the needs of Ontario Works Participants residing in London and Middlesex County. The Ontario Works caseload in London has fluctuated in recent years, and although recently there have been positive downward trends, the caseload still remains as the highest per capita in the Province.

The Ministry of Labour, Training and Skills Development (MLTSD), through the Skills Advance Ontario (SAO), currently funds sector-focused partnerships that connect employers with the employment and training services required to recruit, retain and advance workers with the right essential, technical and employability skills. Jobseekers are provided with sector-specific employment and training services along with coaching and support to obtain and maintain employment. The Skills Advance Ontario (SAO) program with 100% funding from the Ministry of Labour, Training and Skills Development (MLTSD) would augment the current employment services available to Ontario Works Participants.

The City of London has held encouraging discussions with the MLTSD and external stakeholders resulting in the submission of a proposal where The City of London would coordinate the curriculum development and service delivery in London-Middlesex. At an operational level, external

stakeholders include social services, economic development offices and workforce development partners. The Skill Advance Ontario (SAO) project is not a mandatory program which the City is legislatively required to offer; the Skill Advance Ontario (SAO) is an opportunity to provide intensive supports to individuals who require extra supports in attaining and maintaining employment.

Skills Advance Ontario (SAO) Project Summary

The Skills Advance Ontario (SAO) project in London-Middlesex will focus on the specific sector needs of food manufacturing and will provide job seekers with right essential, technical and employability skills identified by employers as the critical skills to successful employment. The SAO project will recruit individuals not be currently participating in the labour market, namely those in receipt of social assistance, who would benefit greatly from the intensive supports in attaining and retaining employment. The SAO project addresses labour shortages in London-Middlesex by linking employers to a supply of job-ready participants through a unique training and skills development program.

One of the key elements of the proposed program is in-class training sessions which focus on preemployment, technical and essential skills. The sessions will be based on curriculum that has been co-designed with employers and will include certifications, experiential learning, and interactions with employers, and worksite visits for program participants. The training is followed by an employment placement period which includes several weeks of intensive supports from program staff for participants and employers. The individualized support for the program participants will continue for a 12-month period.

The Skills Advance Ontario (SAO) project helps prepare individuals for the workplace in unique ways through experiential learning and by providing learning opportunities to address work-life challenges before joining the workforce. Examples include: adjusting the workshop times to mimic shift times, preparing healthy lunches and progressively spending more time standing – all beginning to prepare individuals to be successful in a food manufacturing setting.

Significant and intensive supports are provided by dedicated program staff who help participants remain engaged in the program, problem solve around issues and challenges they face as they prepare for work and as they transition into the workplace. At the same time, the program staff will support the employer including HR, supervisors, and workplace mentors with the successful onboarding and employee retention strategies.

Service Delivery

The City of London will provide the necessary backbone support for the employment services and workplace supports provided in London-Middlesex. The Skills Advance Ontario (SAO) Project would qualify as a Targeted Initiative as outlined in the Ontario Works Employment Services Framework 2019 to 2024 and would augment and leverage existing employment services. (December 10, 2018 CPSC Report - Purchase of Service Agreements – Ontario Works Employment Assistance Services).

The framework recognizes the importance of linkages to employers through multiple channels and through a diversity of service providers; community non-profits, education systems and private sector for-profit businesses. The framework builds on community expertise and collaborative efforts through the Employment Sector Council, Local Employment Planning Council (LEPC), London Economic Development Corporation and the London & Middlesex Local Immigration Partnership, in responding to local labour market needs.

The City of London, with assistance, guidance and expertise from an Advisory Committee will develop and issue two Request for Proposals (RFP); there are several organizations in London-Middlesex capable of developing and delivering curriculum as well as a number of employment service providers well positioned to deliver the needed services. The two RFP would be for the following services:

1) Curriculum Development and Delivery of Essential and Technical Training

The Successful Proponent will be responsible for the development of manufacturing sectorspecific curriculum as well as the coordination and delivery of the essential and technical training for each cohort. The Successful Proponent will tailor sector specific training to match the needs of participating employers in London-Middlesex.

2) Service Delivery – Client and Employer

The Successful Proponent will be responsible for the delivery and coordination of client services including recruitment, orientation, pre-employment services, facilitation, job placement services and ongoing coaching and supports to clients. The Successful Proponent will also provide direct on-site supports and coaching to the employers and their staff

The City of London will hire a Program Manager to lead and coordinate the project; working directly with Curriculum Development Partner, the Service Delivery Partner(s), to fully engage the Advisory Committee, and to ensure the timely flow of information including direct communication with the Ministry to share updates, statistics and adhoc reports.

The Social Services departments in the region will play a key role in identifying potential suitable and eligible Ontario Works client, providing referrals and leveraging other supports such as assistance with employment related expenses. The Skills Advance Ontario (SAO) project, when fully operationalized, will graduate an approximately 60 Participants per year. The SAO project will assist in addressing current labour market demands and recruitment challenges experienced by local employers.

Risk Management has reviewed and has noted the Funding Agreement standard terms and conditions that are not subject to change. Therefore the following risk is identified:

The City is required to indemnify and hold the Province of Ontario harmless including its officers, employees, directors, independent contractors, subcontractors, agents, successors and assigns and Her Majesty the Queen in right of Ontario, her ministers, agents, appointees, employees and authorized subcontractors under Schedule A, Section 11.1 on behalf of the Province of Ontario. Such indemnification is potentially limitless, and is set out below:

The Recipient hereby agrees to indemnify and hold harmless the Indemnified Parties from and against any and all liability, loss, costs, damages and expenses (including legal, expert and consultant fees), causes of action, actions, claims, demands, lawsuits or other proceedings, by whomever made, sustained, incurred, brought or prosecuted, in any way arising out of or in connection with the Project or otherwise in connection with the Agreement, unless solely caused by the negligence or wilful misconduct of the Province.

While this provision exposes the City to liability, it should not outweigh the benefits of the funding and the City will mitigate the associated risks by using the optimum level of oversight, control and discipline to ensure that any 3rd party delivery agent and Sub-Project Agreement Holder(s) meet the City's objectives. This will be done by using clearly defined expectations of the objectives, functions, eligibility, criteria, and recipient obligations as set out clearly in the RFP as well as in the resulting Purchase of Service agreements with the Successful Proponents.

The terms of the Ontario Transfer Payment Agreement – Skills Advance Ontario (SAO) as presented in Schedule 1 will not change. The agreement specifics outlined in Schedule B – Project Specific Information, Schedule C – Timelines, Schedule D – Budget and G – Performance will be updated and provided to Council on March 24th as the agreement is currently being prepared by the Ministry.

FINANCIAL IMPACT

This Agreement is 100% Funded by the Ministry of Labour, Training and Skills Development and therefore there is no financial impact to the City of London. The Funding Agreement is for the period March 24, 2020 to March 31, 2021.

CONCLUSION

Increasing both labour market participation and the labour pool will help address a labour market imbalance. The Skills Advance Ontario (SAO) Project in London-Middlesex will help to alleviate the labour market pressures felt by local employers who are currently hiring and planning for further workforce expansion. The Skills Advance Ontario (SAO) Project provides a unique opportunity to assist individuals in moving from social assistance into sustainable employment.

PREPARED BY:	SUBMITTED BY:
JACK SMIT MANAGER, EMPLOYMENT STRATEGIES	KEVIN DICKINS MANAGER, EMPLOYMENT AND INCOME SUPPORT SERVICES
RECOMMENDED BY:	
SANDRA DATARS BERE MANAGING DIRECTOR, HOUSING, SOCIAL SERVICES AND DEARNESS HOME	

Cc: Elaine Sauve, Program Supervisor, Ministry of Community and Social Services Mark Harrison, Regional Program Manager (Acting), Employment and Training Division, Western Region, Ministry of Labour, Training and Skills Development Anna Lisa Barbon, Managing Director, Corporate Services & City Treasurer, Chief

Financial Officer
Bryan Baar, Senior Financial Business Administrator

Annette Ripepi, Financial Business Administrator

Lynn Marshall, Solicitor II, City Solicitor's Office

John Freeman, Manager, Purchasing & Supply

Robert Collins, Director of Workforce Development, LEDC

Deb Mountenay, Executive Director, Elgin Middlesex Oxford Workforce Planning and Development Board

Gerry Macartney, CEO, London Chamber of Commerce

Cara Finn, Director of Economic Development, County of Middlesex

Cindy Howard, GM, Finance and Community Services, County of Middlesex

APPENDIX A

Bill No. 2020

By-law No.

A By-law to approve the Ontario Transfer Payment Agreement – Skills Advance Ontario (SAO) - Employment Services for the Manufacturing Sector in London - Middlesex between Her Majesty the Queen in Right of Ontario as represented by the Minister of Labour, Training and Skills Development and The Corporation of the City of London.

WHEREAS section 2 of the *Municipal Act, 2001*, S.O. 2001, c.25, as amended, provides that municipalities are created by the Province of Ontario to be responsible and accountable governments with respect to matters within their jurisdiction and each municipality is given powers and duties under this Act and many other Acts for the purpose of providing good government with respect to those matters;

AND WHEREAS subsection 5(3) of the *Municipal Act, 2001* provides that a municipal power shall be exercised by by-law;

AND WHEREAS section 9 of the *Municipal Act, 2001* provides that a municipality has the capacity, rights, powers and privileges of a natural person for the purpose of exercising its authority under this or any other Act;

AND WHEREAS section 10 of the *Municipal Act, 2001* provides that the City may provide any service or thing that the City considers necessary or desirable for the public, and may pass by-laws respecting same, and respecting economic, social and environmental well-being of the City, and the health, safety and well-being of persons;

NOW THEREFORE the Municipal Council of The Corporation of the City of London enacts as follows:

- The Ontario Transfer Payment Agreement Skills Advance Ontario (SAO) Employment Services for the Manufacturing Sector in London–Middlesex
 between Her Majesty the Queen in Right of Ontario as represented by the Minister
 of Labour, Training and Skills Development and The Corporation of the City of
 London, substantially in the form attached as Schedule 1 to this by-law, is
 authorized and approved.
- 2. The Mayor and City Clerk are authorized to execute the Ontario Transfer Payment Agreement Skills Advance Ontario (SAO) Employment Services for the Manufacturing Sector in London-Middlesex approved in subsection 1, above.
- The Managing Director, Housing, Social Services and Dearness Home is delegated the authority to approve any further Amendments to Ontario Transfer

Payment Agreement – Skills Advance Ontario (SAO) - Employment Services for the Manufacturing Sector in London–Middlesex if the Amendments are substantially in the form of the Agreement approved in subsection 1, above.

- 4. The Managing Director Housing, Social Services and Dearness Home, or written designate, are authorized to execute any Amendments to the Ontario Transfer Payment Agreement – Skills Advance Ontario (SAO) - Employment Services for the Manufacturing Sector in London–Middlesex approved in subsection 1, above.
- 5. The Managing Director, Housing, Social Services and Dearness Home, or written designate, is delegated the authority to undertake all the administrative, financial and reporting acts, including signing authority regarding: application forms for funding, budgets, cash flows, other financial reporting including financial claims, and directions, consents and other authorizations as may be required, provided that the monetary amounts do not exceed the maximum amount of the Ministry of Labour, Training and Skills Development's contribution specified in the Agreement that are necessary in connection with the Ontario Transfer Payment Agreement Skills Advance Ontario (SAO) Employment Services for the Manufacturing Sector in London-Middlesex approved in subsection 1, above.
- 6. This by-law shall come into force and effect on the day it is passed.

PASSED in Open Council

, 2020.

Ed Holder Mayor

Catharine Saunders City Clerk

First reading -Second reading -Third reading -

ONTARIO TRANSFER PAYMENT AGREEMENT SKILLSADVANCE ONTARIO (SAO)

THE AGREEMENT	, effective as of the day of, 20 [ENTER the
	e agreement] (the "Effective Date")
BETWEEN:	
	Her Majesty the Queen in right of Ontario as represented by the Minister of Labour, Training and Skills Development
	(the "Province")
	- and -
	[ENTER the full legal name of the Recipient]
	(the "Recipient")

BACKGROUND

The Recipient intends to carry out the Project.

The Province wishes to provide Funds to the Recipient for the Project.

CONSIDERATION

In consideration of the mutual covenants and agreements contained in this Agreement and for other good and valuable consideration, the receipt and sufficiency of which is expressly acknowledged, the Province and the Recipient (the "Parties") agree as follows:

ENTIRE AGREEMENT

This agreement (the "Agreement"), including:

Schedule "A" -General Terms and Conditions

Schedule "B" -Project Specific Information and Additional Provisions

Schedule "C" -**Project Description and Timelines**

Schedule "D" -Budget

Schedule "E" -Payment Plan Schedule "F" -Reporting

Schedule "G" -**Performance Commitments** Schedule "H" - Audit and Accountability Requirements; and any amending agreement entered into as provided for below,

constitutes the entire agreement between the Parties with respect to the subject matter contained in this Agreement and supersedes all prior oral or written representations and agreements.

COUNTERPARTS

The Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute one and the same instrument.

AMENDING THE AGREEMENT

The Agreement may only be amended by a written agreement duly executed by the Parties.

ACKNOWLEDGEMENT

The Recipient:

- a. acknowledges that it has read and understands the provisions contained in the entire Agreement; and
- b. agrees to be bound by the terms and conditions contained in the entire Agreement.

IN WITNESS WHEREOF, the Parties have executed the Agreement on the dates set out below.

	HER MAJESTY THE QUEEN IN RIGHT OF ONTARIO as represented by the Minister of Labour, Training and Skills Development
	by:
[ENTER date]	
Date	Name: [ENTER the full legal name of the Ministry officer]
	Title: [ENTER the title of the authorized signing officer]
	Authorized Signing Officer
	[ENTER the full legal name of the Recipient]
	by:
Date	Name: [ENTER the full legal name of Recipient's officer]
	Title: [ENTER the title of the Recipient's 1st signing officer]
	by:
Date	Name: [ENTER full legal name of the Recipient's officer]
	Title: [ENTER the title of the Recipient's 2 nd signing officer]
	I/We have authority to bind the Recipient.

SCHEDULE "A"

GENERAL TERMS AND CONDITIONS

1.0 INTERPRETATION AND DEFINITIONS

- 1.1 **Interpretation.** For the purposes of interpretation:
 - a. words in the singular include the plural and vice-versa;
 - b. words in one gender include all genders;
 - c. the headings do not form part of the Agreement; they are for reference only and will not affect the interpretation of the Agreement;
 - d. any reference to dollars or currency will be in Canadian dollars and currency; and
 - e. "include", "includes" and "including" denote that the subsequent list is not exhaustive.
- 1.2 **Definitions**. In the Agreement, the following terms will have the following meanings:
 - "Additional Provisions" means the terms and conditions referred to in section 9.1 and as specified in Schedule "B".
 - "BPSAA" means the Broader Public Sector Accountability Act, 2010 (Ontario).
 - "Budget" means the budget attached to the Agreement as Schedule "D".
 - "Business Day" means any working day, Monday to Friday inclusive, excluding statutory and other holidays, namely: New Year's Day; Family Day; Good Friday; Easter Monday; Victoria Day; Canada Day; Civic Holiday; Labour Day; Thanksgiving Day; Remembrance Day; Christmas Day; Boxing Day and any other day on which the Province has elected to be closed for business.
 - "Event of Default" has the meaning ascribed to it in section15.1.
 - "Expiration Date" means the date on which this Agreement will expire and is the date set out in Schedule "B".
 - "Funding Year" means:
 - a. in the case of the first Funding Year, the period commencing on the Effective

- Date and ending on the following March 31; and
- b. in the case of Funding Years subsequent to the first Funding Year, the period commencing on April 1 following the end of the previous Funding Year and ending on the following March 31.
- **"Funds"** means the money the Province provides to the Recipient pursuant to the Agreement.
- "Indemnified Parties" means Her Majesty the Queen in right of Ontario, Her ministers, agents, appointees and employees.
- "Maximum Funds" means the maximum amount the Province will provide the Recipient under the Agreement as set out in Schedule "B".
- "Notice" means any communication given or required to be given pursuant to the Agreement.
- "Notice Period" means the period of time within which the Recipient is required to remedy an Event of Default, and includes any such period or periods of time by which the Province considers it reasonable to extend that time.
- "Party" means either the Province or the Recipient.
- "Project" means the undertaking described in Schedule "C".
- "PSSDA" means the Public Sector Salary Disclosure Act, 1996 (Ontario).
- "Reports" means the reports described in Schedule "F".
- "Timelines" means the Project schedule set out in Schedule "C".

2.0 REPRESENTATIONS, WARRANTIES AND COVENANTS

- 2.1 **General.** The Recipient represents, warrants and covenants that:
 - a. it is, and will continue to be for the term of the Agreement, a validly existing legal entity with full power to fulfill its obligations under the Agreement;
 - b. it has, and will continue to have for the term of the Agreement, the experience and expertise necessary to carry out the Project;
 - c. it is in compliance with all federal and provincial laws and regulations, all municipal by-laws, and any other orders, rules and by-laws related to any aspect of the Project, the Funds or both; and

- d. unless otherwise provided for in the Agreement, any information the Recipient provided to the Province in support of its request for funds (including information relating to any eligibility requirements) was true and complete at the time the Recipient provided it and will continue to be true and complete for the term of the Agreement.
- 2.2 **Execution of Agreement.** The Recipient represents and warrants that it has:
 - a. the full power and authority to enter into the Agreement; and
 - b. taken all necessary actions to authorize the execution of the Agreement.
- 2.3 **Governance.** The Recipient represents, warrants and covenants that it has, and will maintain, in writing for the period during which the Agreement is in effect:
 - a. a code of conduct and ethical responsibilities for all persons at all levels of the Recipient's organization;
 - b. procedures to ensure the ongoing effective functioning of the Recipient;
 - c. decision-making mechanisms for the Recipient;
 - d. procedures to enable the Recipient to manage Funds prudently and effectively;
 - e. procedures to enable the Recipient to complete the Project successfully;
 - f. procedures to enable the Recipient, in a timely manner, to identify risks to the completion of the Project, and strategies to address the identified risks;
 - g. procedures to enable the preparation and delivery of all Reports required pursuant to Article 7.0; and
 - h. procedures to enable the Recipient to deal with such other matters as the Recipient considers necessary to ensure that the Recipient carries out its obligations under the Agreement.
- 2.4 **Supporting Documentation.** Upon request, the Recipient will provide the Province with proof of the matters referred to in this Article 2.0.

3.0 TERM OF THE AGREEMENT

3.1 **Term.** The term of the Agreement will commence on the Effective Date and will expire on the Expiration Date unless terminated earlier pursuant to Article 13.0, Article 14.0 or Article 15.0.

4.0 FUNDS AND CARRYING OUT THE PROJECT

4.1 **Funds Provided**. The Province will:

- a. provide the Recipient up to the Maximum Funds for the purpose of carrying out the Project;
- b. provide the Funds to the Recipient in accordance with the Payment Plan attached to the Agreement as Schedule "E"; and
- c. deposit the Funds into an account designated by the Recipient provided that the account:
 - i. resides at a Canadian financial institution; and
 - ii. is in the name of the Recipient.

4.2 **Limitation on Payment of Funds**. Despite section 4.1:

- a. the Province is not obligated to provide any Funds to the Recipient until the Recipient provides the insurance certificate or other proof as the Province may request pursuant to section 12.2;
- b. the Province is not obligated to provide instalments of Funds until it is satisfied with the progress of the Project;
- c. the Province may adjust the amount of Funds it provides to the Recipient in any Funding Year based upon the Province's assessment of the information provided by the Recipient pursuant to section 7.1; and
- d. if, pursuant to the *Financial Administration Act* (Ontario), the Province does not receive the necessary appropriation from the Ontario Legislature for payment under the Agreement, the Province is not obligated to make any such payment, and, as a consequence, the Province may:
 - reduce the amount of Funds and, in consultation with the Recipient, change the Project; or
 - ii. terminate the Agreement pursuant to section 14.1.

4.3 Use of Funds and Project. The Recipient will:

a. carry out the Project in accordance with the terms and conditions of the Agreement;

- b. use the Funds only for the purpose of carrying out the Project;
- c. spend the Funds only in accordance with the Budget; and
- d. not use the Funds to cover any specific cost that has or will be funded or reimbursed by any third party, including other ministries, agencies and organizations of the Government of Ontario.
- 4.4 **Province's Role Limited to Providing Funds**. For greater clarity, the Province's role under the Agreement is limited to providing Funds to the Recipient for the purposes of the Project and the Province is not responsible for carrying out the Project.
- 4.5 **No Changes**. The Recipient will not make any changes to the Project, the Timelines, or the Budget without the prior written consent of the Province.
- 4.6 **Interest Bearing Account**. If the Province provides Funds to the Recipient before the Recipient's immediate need for the Funds, the Recipient will place the Funds in an interest bearing account in the name of the Recipient at a Canadian financial institution.
- 4.7 **Interest**. If the Recipient earns any interest on the Funds, the Province may:
 - a. deduct an amount equal to the interest from any further instalments of Funds;
 or
 - b. demand from the Recipient the repayment of an amount equal to the interest.
- 4.8 **Maximum Funds**. The Recipient acknowledges that the Funds available to it pursuant to the Agreement will not exceed the Maximum Funds.
- 4.9 **Rebates, Credits and Refunds**. The Recipient acknowledges that the amount of Funds available to it pursuant to the Agreement is based on the actual costs to the Recipient, less any costs (including taxes) for which the Recipient has received, will receive, or is eligible to receive, a rebate, credit or refund.
- 4.10 **Funding, Not Procurement**. For greater clarity, the Recipient acknowledges that:
 - a. it is receiving funding from the Province for the Project and is not providing goods or services to the Province; and
 - b. the funding the Province is providing under the Agreement is funding for the purposes of the PSSDA.
- 5.0 RECIPIENT'S ACQUISITION OF GOODS OR SERVICES, AND DISPOSAL OF

ASSETS

- 5.1 **Acquisition**. If the Recipient acquires goods, services, or both with the Funds, it will:
 - a. do so through a process that promotes the best value for money; and
 - b. comply with the BPSAA, including any procurement directive issued thereunder, to the extent applicable.
- 5.2 **Disposal**. The Recipient will not, without the Province's prior written consent, sell, lease or otherwise dispose of any asset purchased or created with the Funds or for which Funds were provided, the cost of which exceeded the amount as set out in Schedule "B" at the time of purchase.

6.0 CONFLICT OF INTEREST

- 6.1 **No Conflict of Interest**. The Recipient will carry out the Project and use the Funds without an actual, potential or perceived conflict of interest.
- 6.2 **Conflict of Interest Includes**. For the purposes of this Article, a conflict of interest includes any circumstances where:
 - a. the Recipient; or
 - b. any person who has the capacity to influence the Recipient's decisions,

has outside commitments, relationships or financial interests that could, or could be seen to, interfere with the Recipient's objective, unbiased and impartial judgment relating to the Project, the use of the Funds, or both.

- 6.3 **Disclosure to Province**. The Recipient will:
 - a. disclose to the Province, without delay, any situation that a reasonable person would interpret as an actual, potential or perceived conflict of interest; and
 - b. comply with any terms and conditions that the Province may prescribe as a result of the disclosure.

7.0 REPORTING, ACCOUNTING AND REVIEW

7.1 **Preparation and Submission**. The Recipient will:

- a. submit to the Province at the address referred to in section 19.1, all Reports in accordance with the timelines and content requirements set out in Schedule "F", or in a form as specified by the Province from time to time;
- b. submit to the Province at the address referred to in section19.1, any other reports as may be requested by the Province in accordance with the timelines and content requirements specified by the Province;
- c. ensure that all Reports and other reports are completed to the satisfaction of the Province; and
- d. ensure that all Reports and other reports are signed on behalf of the Recipient by an authorized signing officer.
- 7.2 **Record Maintenance**. The Recipient will keep and maintain:
 - all financial records (including invoices) relating to the Funds or otherwise to the Project in a manner consistent with generally accepted accounting principles; and
 - b. all non-financial documents and records relating to the Funds or otherwise to the Project.
- 7.3 **Inspection**. The Province, its authorized representatives or an independent auditor identified by the Province may, at its own expense, upon twenty-four hours' Notice to the Recipient and during normal business hours, enter upon the Recipient's premises to review the progress of the Project and the Recipient's allocation and expenditure of the Funds and, for these purposes, the Province, its authorized representatives or an independent auditor identified by the Province may take one or more of the following actions:
 - a. inspect and copy the records and documents referred to in section 7.2;
 - b. remove any copies made pursuant to section 7.3a. from the Recipient's premises; and
 - c. conduct an audit or investigation of the Recipient in respect of the expenditure of the Funds, the Project, or both.
- 7.4 **Disclosure**. To assist in respect of the rights set out in section 7.3, the Recipient will disclose any information requested by the Province, its authorized representatives or an independent auditor identified by the Province, and will do so in the form requested by the Province, its authorized representatives or an independent auditor identified by the Province, as the case may be.

- 7.5 **No Control of Records**. No provision of the Agreement will be construed so as to give the Province any control whatsoever over the Recipient's records.
- 7.6 **Auditor General**. For greater certainty, the Province's rights under this Article are in addition to any rights provided to the Auditor General pursuant to section 9.1 of the *Auditor General Act* (Ontario).

8.0 COMMUNICATIONS REQUIREMENTS

- 8.1 **Acknowledge Support**. Unless otherwise directed by the Province, the Recipient will acknowledge the support of the Province in a form and manner as directed by the Province.
- 8.2 **Publication**. The Recipient will indicate, in any of its Project-related publications, whether written, oral, or visual, that the views expressed in the publication are the views of the Recipient and do not necessarily reflect those of the Province.

9.0 FURTHER CONDITIONS

9.1 **Additional Provisions**. The Recipient will comply with any Additional Provisions. In the event of a conflict or inconsistency between any of the requirements of the Additional Provisions and any requirements of this Schedule "A", the Additional Provisions will prevail.

10.0 FREEDOM OF INFORMATION AND PROTECTION OF PRIVACY

10.1 **FIPPA**. The Recipient acknowledges that the Province is bound by the *Freedom of Information and Protection of Privacy Act* (Ontario) and that any information provided to the Province in connection with the Project or otherwise in connection with the Agreement may be subject to disclosure in accordance with that Act.

11.0 INDEMNITY

11.1 Indemnification. The Recipient hereby agrees to indemnify and hold harmless the Indemnified Parties from and against any and all liability, loss, costs, damages and expenses (including legal, expert and consultant fees), causes of action, actions, claims, demands, lawsuits or other proceedings, by whomever made, sustained, incurred, brought or prosecuted, in any way arising out of or in connection with the Project or otherwise in connection with the Agreement, unless solely caused by the negligence or wilful misconduct of the Province.

- 11.2 **Recipient's Participation**. The Recipient will, at its expense, to the extent requested by the Province, participate in or conduct the defence of any proceeding against any Indemnified Parties and any negotiations for their settlement.
- 11.3 **Province's Election**. The Province may elect to participate in or conduct the defence of any proceeding by providing Notice to the Recipient of such election without prejudice to any other rights or remedies of Province under the Agreement, at law or in equity. Each Party participating in the defence will do so by actively participating with the other's counsel.
- 11.4 **Settlement Authority**. The Recipient will not enter into a settlement of any proceeding against any Indemnified Parties unless the Recipient has obtained the prior written approval of Province. If the Recipient is requested by the Province to participate in or conduct the defence of any proceeding, the Province will co-operate with and assist the Recipient to the fullest extent possible in the proceeding and any related settlement negotiations.
- 11.5 **Recipient's Co-operation**. If the Province conducts the defence of any proceedings, the Recipient will co-operate with and assist the Province to the fullest extent possible in the proceedings and any related settlement negotiations.

12.0 INSURANCE

- 12.1 **Recipient's Insurance**. The Recipient represents and warrants that it has, and will maintain for the term of the Agreement, at its own cost and expense, with insurers having a secure A.M. Best rating of B+ or greater, or the equivalent, all the necessary and appropriate insurance that a prudent person carrying out a project similar to the Project would maintain, including commercial general liability insurance on an occurrence basis for third party bodily injury, personal injury and property damage, to an inclusive limit of not less than the amount set out in Schedule "B" per occurrence. The policy will include the following:
 - a. the Indemnified Parties as additional insureds with respect to liability arising in the course of performance of the Recipient's obligations under, or otherwise in connection with, the Agreement;
 - b. a cross-liability clause;
 - c. contractual liability coverage; and
 - d. a 30 day written notice of cancellation.
- 12.2 **Proof of Insurance**. The Recipient will provide the Province with certificates of insurance, or other proof as may be requested by the Province, that confirms the insurance coverage as provided for in section 12.1. Upon the request of the

Province, the Recipient will make available to the Province a copy of each insurance policy.

13.0 TERMINATION ON NOTICE

- 13.1 **Termination on Notice**. The Province may terminate the Agreement at any time without liability, penalty or costs upon giving at least 30 days' Notice to the Recipient.
- 13.2 **Consequences of Termination on Notice by the Province**. If the Province terminates the Agreement pursuant to section 13.1, the Province may take one or more of the following actions:
 - a. cancel all further instalments of Funds;
 - b. demand the repayment of any Funds remaining in the possession or under the control of the Recipient; and
 - c. determine the reasonable costs for the Recipient to wind down the Project, and do either or both of the following:
 - i. permit the Recipient to offset such costs against the amount owing pursuant to section 13.2b.; and
 - ii. subject to section 4.8, provide Funds to the Recipient to cover such costs.

14.0 TERMINATION WHERE NO APPROPRIATION

- 14.1 **Termination Where No Appropriation**. If, as provided for in section 4.2d., the Province does not receive the necessary appropriation from the Ontario Legislature for any payment the Province is to make pursuant to the Agreement, the Province may terminate the Agreement immediately without liability, penalty or costs by giving Notice to the Recipient.
- 14.2 **Consequences of Termination Where No Appropriation**. If the Province terminates the Agreement pursuant to section 14.1, the Province may take one or more of the following actions:
 - a. cancel all further instalments of Funds;
 - b. demand the repayment of any Funds remaining in the possession or under the control of the Recipient; and

- c. determine the reasonable costs for the Recipient to wind down the Project and permit the Recipient to offset such costs against the amount owing pursuant to section 14.2b.
- 14.3 **No Additional Funds**. For greater clarity, if the costs determined pursuant to section 14.2c. exceed the Funds remaining in the possession or under the control of the Recipient, the Province will not provide additional Funds to the Recipient.

15.0 EVENT OF DEFAULT, CORRECTIVE ACTION AND TERMINATION FOR DEFAULT

- 15.1 **Events of Default**. Each of the following events will constitute an Event of Default:
 - a. in the opinion of the Province, the Recipient breaches any representation, warranty, covenant or other material term of the Agreement, including failing to do any of the following in accordance with the terms and conditions of the Agreement:
 - i. carry out the Project;
 - ii. use or spend Funds; or
 - iii. provide, in accordance with section 7.1, Reports or such other reports as may have been requested pursuant to section 7.1b.;
 - b. the Recipient's operations, or its organizational structure, changes such that it no longer meets one or more of the eligibility requirements of the program under which the Province provides the Funds;
 - c. the Recipient makes an assignment, proposal, compromise, or arrangement for the benefit of creditors, or a creditor makes an application for an order adjudging the Recipient bankrupt, or applies for the appointment of a receiver; or
 - d. the Recipient ceases to operate.
- 15.2 **Consequences of Events of Default and Corrective Action**. If an Event of Default occurs, the Province may, at any time, take one or more of the following actions:
 - a. initiate any action the Province considers necessary in order to facilitate the successful continuation or completion of the Project;
 - b. provide the Recipient with an opportunity to remedy the Event of Default;

- c. suspend the payment of Funds for such period as the Province determines appropriate;
- d. reduce the amount of the Funds;
- e. cancel all further instalments of Funds;
- f. demand the repayment of any Funds remaining in the possession or under the control of the Recipient;
- g. demand the repayment of an amount equal to any Funds the Recipient used, but did not use in accordance with the Agreement;
- h. demand the repayment of an amount equal to any Funds the Province provided to the Recipient; and
- i. terminate the Agreement at any time, including immediately, without liability, penalty or costs to the Province upon giving Notice to the Recipient.
- 15.3 **Opportunity to Remedy**. If, in accordance with section 15.2b., the Province provides the Recipient with an opportunity to remedy the Event of Default, the Province will provide Notice to the Recipient of:
 - a. the particulars of the Event of Default; and
 - b. the Notice Period.
- 15.4 **Recipient not Remedying**. If the Province has provided the Recipient with an opportunity to remedy the Event of Default pursuant to section 15.2b., and:
 - a. the Recipient does not remedy the Event of Default within the Notice Period;
 - b. it becomes apparent to the Province that the Recipient cannot completely remedy the Event of Default within the Notice Period; or
 - c. the Recipient is not proceeding to remedy the Event of Default in a way that is satisfactory to the Province,

the Province may extend the Notice Period, or initiate any one or more of the actions provided for in sections 15.2a., c., d., e., f., g., h. and i.

15.5 **When Termination Effective**. Termination under this Article will take effect as set out in the Notice.

16.0 FUNDS AT THE END OF A FUNDING YEAR

- 16.1 **Funds at the End of a Funding Year**. Without limiting any rights of the Province under Article 15.0, if the Recipient has not spent all of the Funds allocated for the Funding Year as provided for in the Budget, the Province may take one or both of the following actions:
 - a. demand the return of the unspent Funds; and
 - b. adjust the amount of any further instalments of Funds accordingly.

17.0 FUNDS UPON EXPIRY

17.1 **Funds Upon Expiry**. The Recipient will, upon expiry of the Agreement, return to the Province any Funds remaining in its possession or under its control.

18.0 REPAYMENT

- 18.1 **Repayment of Overpayment**. If at any time during the term of the Agreement the Province provides Funds in excess of the amount to which the Recipient is entitled under the Agreement, the Province may:
 - a. deduct an amount equal to the excess Funds from any further instalments of Funds: or
 - b. demand that the Recipient pay an amount equal to the excess Funds to the Province.
- 18.2 **Debt Due**. If, pursuant to the Agreement:
 - a. the Province demands the payment of any Funds or an amount equal to any Funds from the Recipient; or
 - b. the Recipient owes any Funds or an amount equal to any Funds to the Province, whether or not their return or repayment has been demanded by the Province.

such Funds or other amount will be deemed to be a debt due and owing to the Province by the Recipient, and the Recipient will pay or return the amount to the Province immediately, unless the Province directs otherwise.

18.3 **Interest Rate**. The Province may charge the Recipient interest on any money owing by the Recipient at the then current interest rate charged by the Province of Ontario on accounts receivable.

- 18.4 **Payment of Money to Province**. The Recipient will pay any money owing to the Province by cheque payable to the "Ontario Minister of Finance" and delivered to the Province at the address referred to in section 19.1.
- 18.5 **Failure to Repay**. Without limiting the application of section 43 of the *Financial Administration Act* (Ontario), if the Recipient fails to repay any amount owing under the Agreement, Her Majesty the Queen in right of Ontario may deduct any unpaid amount from any money payable to the Recipient by Her Majesty the Queen in right of Ontario.

19.0 NOTICE

- 19.1 **Notice in Writing and Addressed**. Notice will be in writing and will be delivered by email, postage-prepaid mail, personal delivery or fax, and will be addressed to the Province and the Recipient respectively as set out in Schedule "B", or as either Party later designates to the other by Notice.
- 19.2 **Notice Given**. Notice will be deemed to have been given:
 - a. in the case of postage-prepaid mail, five Business Days after the Notice is mailed; or
 - b. in the case of email, personal delivery or fax, one Business Day after the Notice is delivered.
- 19.3 **Postal Disruption**. Despite section 19.2a., in the event of a postal disruption:
 - a. Notice by postage-prepaid mail will not be deemed to be received; and
 - b. the Party giving Notice will provide Notice by email, personal delivery or by fax.

20.0 CONSENT BY PROVINCE AND COMPLIANCE BY RECIPIENT

20.1 Consent. When the Province provides its consent pursuant to the Agreement, it may impose any terms and conditions on such consent and the Recipient will comply with such terms and conditions.

21.0 SEVERABILITY OF PROVISIONS

21.1 **Invalidity or Unenforceability of Any Provision**. The invalidity or unenforceability of any provision of the Agreement will not affect the validity or enforceability of any

other provision of the Agreement. Any invalid or unenforceable provision will be deemed to be severed.

22.0 WAIVER

22.1 **Waivers in Writing**. If a Party fails to comply with any term of the Agreement, that Party may only rely on a waiver of the other Party if the other Party has provided a written waiver in accordance with the Notice provisions in Article 19.0. Any waiver must refer to a specific failure to comply and will not have the effect of waiving any subsequent failures to comply.

23.0 INDEPENDENT PARTIES

23.1 **Parties Independent**. The Recipient acknowledges that it is not an agent, joint venturer, partner or employee of the Province, and the Recipient will not represent itself in any way that might be taken by a reasonable person to suggest that it is, or take any actions that could establish or imply such a relationship.

24.0 ASSIGNMENT OF AGREEMENT OR FUNDS

- 24.1 **No Assignment**. The Recipient will not, without the prior written consent of the Province, assign any of its rights, or obligations under the Agreement.
- 24.2 **Agreement Binding**. All rights and obligations contained in the Agreement will extend to and be binding on the Parties' respective heirs, executors, administrators, successors and permitted assigns.

25.0 GOVERNING LAW

25.1 **Governing Law**. The Agreement and the rights, obligations and relations of the Parties will be governed by and construed in accordance with the laws of the Province of Ontario and the applicable federal laws of Canada. Any actions or proceedings arising in connection with the Agreement will be conducted in the courts of Ontario, which will have exclusive jurisdiction over such proceedings.

26.0 FURTHER ASSURANCES

26.1 **Agreement into Effect**. The Recipient will provide such further assurances as the Province may request from time to time with respect to any matter to which the

Agreement pertains, and will otherwise do or cause to be done all acts or things necessary to implement and carry into effect the terms and conditions of the Agreement to their full extent.

27.0 JOINT AND SEVERAL LIABILITY

27.1 **Joint and Several Liability**. Where the Recipient is comprised of more than one entity, all such entities will be jointly and severally liable to the Province for the fulfillment of the obligations of the Recipient under the Agreement.

28.0 RIGHTS AND REMEDIES CUMULATIVE

28.1 **Rights and Remedies Cumulative**. The rights and remedies of the Province under the Agreement are cumulative and are in addition to, and not in substitution for, any of its rights and remedies provided by law or in equity.

29.0 ACKNOWLEDGEMENT OF OTHER LEGISLATION AND DIRECTIVES

29.1 Recipient Acknowledges. The Recipient:

- a. acknowledges that by receiving Funds it may become subject to legislation applicable to organizations that receive funding from the Government of Ontario, including the BPSAA, the PSSDA, and the *Auditor General Act* (Ontario);
- acknowledges that Her Majesty the Queen in right of Ontario has issued expenses, perquisites, and procurement directives and guidelines pursuant to the BPSAA; and
- c. will comply with any such legislation, including directives issued thereunder, to the extent applicable.

30.0 FAILURE TO COMPLY WITH OTHER AGREEMENTS

30.1 Other Agreements. If the Recipient:

a. has failed to comply (a "Failure") with any term, condition or obligation under any other agreement with Her Majesty the Queen in right of Ontario or one of Her agencies;

- b. has been provided with notice of such Failure in accordance with the requirements of such other agreement;
- c. has, if applicable, failed to rectify such Failure in accordance with the requirements of such other agreement; and
- d. such Failure is continuing,

the Province may suspend the payment of Funds for such period as the Province determines appropriate.

31.0 SURVIVAL

31.1 **Survival**. The following Articles and sections, and all applicable cross-referenced sections and schedules, will continue in full force and effect for a period of seven years from the date of expiry or termination of the Agreement: Article 1.0 and any other applicable definitions, section 4.2d., 4.7, section 5.2, section 7.1 (to the extent that the Recipient has not provided the Reports to the satisfaction of the Province), sections 7.2, 7.3, 7.4, 7.5, 7.6, Article 8.0, Article 11.0, section 13.2, sections 14.2 and 14.3, sections 15.1, 15.2d., e., f., g. and h., Article 17.0, Article 18.0, Article 19.0, Article 21.0, section 24.2, Article 25.0, Article 27.0, Article 28.0, Article 29.0, Article 30.0, and Article 31.0.

- END OF GENERAL TERMS AND CONDITIONS -

SCHEDULE "B"

PROJECT SPECIFIC INFORMATION AND ADDITIONAL PROVISIONS

Maximum Funds	\$ [ENTER maximum funds of agreement]
Expiration Date	[ENTER end date of agreement]
Amount for the purposes of section 5.2 of Schedule "A"	\$1,000
Insurance	\$ 2,000,000
Contact information for the purposes of Notice to the Province	Name: Ministry of Labour, Training and Skills Development Address: [ENTER Ministry address] Attention: [ENTER Ministry contact name]
	Fax: [ENTER Ministry fax number]
	Email: [ENTER Ministry email]
Contact information for the purposes of Notice to the Recipient	Name: [ENTER name of the Recipient organization] Address: [ENTER Recipient's address] Attention: [ENTER Recipient contact name] Fax: [ENTER Recipient fax number] Email: [ENTER Recipient email]
Contact information for the senior financial person in the Recipient organization (e.g., CFO, CAO) to respond as required to requests from the Province related to the Agreement	Name: [ENTER name of the Recipient organization] Position: [ENTER position] Fax: [ENTER fax number] Email: [ENTER email]

Additional Provisions:

1. Amendments to Definitions

The definition of "Indemnified Parties" in Article 1.2 is deleted and replaced with:

"Indemnified Parties" means Her Majesty the Queen in right of Ontario, Her ministers, agents, appointees, employees and authorized subcontractors.

2. Further Definitions

The following definitions also apply to this Agreement

"EOIS-CaMS" means the Employment Ontario Information System-Case Management System.

"Employer" means a person or a firm registered in SAO and actively receiving SAO services; and/or informing the development and delivery of sector-focused employment and training services to ensure that Participants have the right essential, technical, and employability skills to obtain entry-level employment and advance in identified sectors.

"EOPG" means the <u>Employment Ontario Partners' Gateway</u>, a provincially maintained website which provides support to service providers delivering Employment Ontario programs and services as part of the Employment Ontario network.

"FIPPA" means the Freedom of Information and Protection of Privacy Act, R.S.O. 1990, c. F. 31, as amended.

"Incumbent Worker" means a person registered in SAO as a Participant who is already employed, but is either vulnerable in his or her current employment or faces barriers to advancement before receiving SAO services.

"Individuals" means a person who is interested in becoming a Participant.

"Jobseeker" means a person registered in SAO as a Participant and is unemployed before receiving SAO services.

"Participant" means a person registered in SAO and actively receiving SAO services.

"Participant Intake Target" means a Participant who has started receiving SAO services in 2019-2020.

"Participant Placement Target" means a Participant who has started a job placement in 2019-2020.

"Participant Retention Target" means a Participant who has completed their job placement and remains employed with the placement employer.

"SAO" means SkillsAdvance Ontario.

"Sector" means multiple employers or organizations that are in the same or a similar or related industry.

"SPRA" means Service Provider Registration Authority.

3. Additions to Article 2.0

The following provisions are added to Article 2.1:

- e. it has provided, and will continue to provide for the term of the Agreement, training to any person involved in carrying out the Project on the relevant responsibilities of the Recipient set out in the Agreement; and
- f. it has taken, and will continue to take for the term of the Agreement, all reasonable actions to minimize and reduce the costs related to the Project that may be incurred as a result of the expiry or termination of the Agreement including negotiating all contracts related to the Project, such as employment contracts, on terms that will enable the Recipient to cancel them upon terms and conditions that will minimize their cancellation costs in the event of the expiry or termination of the Agreement

4. Additions to Article 7.0

The following provision is added to Article 7.0:

7.7 Records Transfer. At any time after the expiry of the Agreement or the termination of the Agreement pursuant to Article 13.0, Article 14.0 or Article 15.0, the Province may demand the transfer of any records referred to in section 7.2 to another organization identified by the Province in order to facilitate the successful continuation or completion of the Project, or a similar project, including the continuation or completion of services to individuals.

5. Amendment and Addition to Article 8.0

Article 8.2 is deleted and replaced with:

8.2 **Publication**. The Recipient will indicate, in any of its Project-related publications, whether written, oral, or visual, that the views expressed in the publication are the views of the Recipient and do not necessarily reflect those of the

Province, and shall include a statement that "The Government of Ontario and its agencies are in no way bound by the recommendations contained in this document."

The following provision is added to Article 8.0:

8.3 **Visual Identity and Communications.** The Recipient will comply with the Visual Identity and Communications Guidelines for Employment Ontario Services, available on the EOPG, and which may be amended from time to time at the sole discretion of the Province.

6. Additions to Article 10.0

The following provisions are added to Article 10.0:

- 10.2 Idem. The Recipient acknowledges that the Province is bound by FIPPA and undertakes to perform its obligations under the Agreement in a manner that ensures that the Province is not in breach of its obligations under FIPPA.
- 10.3 **Protection of Privacy.** The Recipient represents and warrants that it will protect privacy in accordance with applicable privacy legislation or pursuant to its own privacy policy that is consistent with the Canadian Standards Association Code for the Protection of Personal Information and that is publicly available. Without limitation, the Recipient will:
 - designate an experienced official who will be responsible for ensuring the Recipient's compliance with its privacy policy and the privacy protection provisions of the Agreement;
 - b. make the designated privacy officer aware of the privacy policy and the privacy protection provisions of the Agreement;
 - implement appropriate privacy protection training of employees, contractors and authorized subcontractors who have access to personal information to deliver the Project;
 - only collect, use and disclose personal information if necessary to deliver the Project to the Recipient's Participants and comply with its obligations under the Agreement;
 - e. ensure that the personal information of the Recipient's Participants, including contact information, is accurate and up to date;
 - f. at the earliest opportunity, provide notice of collection, if required,

and obtain and retain any consents that may be necessary to deliver the Project and comply with its obligations under the Agreement in compliance with all applicable privacy legislation, and the Recipient's privacy policy;

- g. on behalf of the Province, provide each Participant with the Notice of Collection of Personal Information and obtain his or her Consent to the indirect collection of personal information set out on the forms as mandated by the Province for this Project and retain the signed forms as mandated by the Province for this Project for a period of seven (7) years and make it available to the Province, upon request;
- not use personal information that was collected for use in delivering the Project to the Recipient's Participants for any other purpose without the informed and voluntary written consent of the individual;
- limit access to personal information to employees, contractors and authorized subcontractors who need the personal information to deliver the Project to the Recipient's Participants and ensure the Recipient's compliance with its obligations under the Agreement;
- j. ensure that only the SPRA and employees authorized by the SPRA have access to the EOIS-CaMS, using their assigned EOIS-CaMS Enrolment Numbers and PIN IDs and that the SPRA and the authorized users abide by the SPRA Terms and Conditions and the representations made by the Recipient on the SPRA EOIS-CaMS Registration form;
- before disclosing personal information to employees, contractors and authorized subcontractors, enter into an agreement with them requiring them to be bound by the Recipient's privacy policy and the privacy protection provisions of the Agreement;
- ensure the security and integrity of any personal information collected by implementing, using and maintaining the most appropriate products, tools, measures and procedures to prevent the unauthorized or inadvertent collection, use, disclosure, loss, alteration or destruction;
- m. provide Participants with access to their own personal information in accordance with applicable privacy legislation, and the Recipient's privacy policy;

- ensure the secure and irreversible destruction of paper records containing personal information when it is no longer needed to deliver the Project or to comply with the obligations under the Agreement;
- ensure that electronic records containing personal information that are no longer needed to deliver the Project or to comply with the obligations under the Agreement are not accessible until secure and irreversible destruction of these records is possible;
- p. notify the Province as soon as the Recipient becomes aware of a potential or actual breach of any of the privacy policy and the privacy protection provisions of the Agreement;
- cooperate with the Province and its contractors and auditors in any audit of or investigation into a breach of the privacy policy and the privacy protection provisions of the Agreement; and
- r. implement, use and maintain other specific privacy or security measures that in the reasonable opinion of the Province would improve the adequacy and effectiveness of the Recipient's measures to ensure the privacy and security of the records collected, created, used and disclosed in the delivery of the Project and compliance with the obligations under the Agreement.

7. Amendment to Article 12.0

Article 12.1 is deleted and replaced with:

- 12.1 **Recipient's Insurance.** The Recipient represents and warrants that it has, and will maintain for the term of the Agreement, at its own cost and expense, with insurers having a secure A.M. Best rating of B+ or greater, or the equivalent, or if the Recipient is a school board, with the Ontario School Boards' Insurance Exchange, all the necessary and appropriate insurance that a prudent person carrying out a project similar to the Project would maintain, including commercial general liability insurance on an occurrence basis for third party bodily injury, personal injury and property damage, to an inclusive limit of not less than two million dollars (\$2,000,000) per occurrence. The policy will include the following:
 - a. the Indemnified Parties as additional insureds with respect to liability arising in the course of performance of the Recipient's

- obligations under, or otherwise in connection with, the Agreement;
- b. a cross-liability clause;
- c. contractual liability coverage; and
- d. a 30 day written notice of cancellation, termination or material change.

8. Addition to Article 15.0

The following provision is added to Article 15.1a.:

iv. meet the performance commitments described in Schedule "G".

9. Amendment to Article 31

The following provisions are added to the list in Article 31: Article 10 and Section 8 e. in Schedule C.

10.New Article 32.0

Schedule "A" is amended by adding a new Article 32.0 **INTELLECTUAL PROPERTY**, as follows:

- 32.1 **Ownership of Products.** The Recipient will own the products.
- 32.2 **Recipient's Grant of Licence.** The Recipient grants the Province and all Stakeholders a perpetual, irrevocable and royalty-free licence to use the Products in Ontario for any purpose except commercial gain. Without limitation, the Province may update, revise, copy, translate or distribute the Products to its Stakeholders.
- 32.3 **Representation and Warranty**. The Recipient represents and warrants that the grant of licences pursuant to section 32.2 will not infringe or induce the infringement of any third party intellectual property rights.
- 32.4 **Acceptance of Terms of Licence.** If at any time, the Recipient is granted a licence for other Products pursuant to an agreement with the Province and another organization to carry out a project with the Province, or a successor Government of Ontario program, the Recipient acknowledges and agrees:
 - a. that the licence for the other Products is solely applicable for the purpose of the Recipient carrying out the Project or a subsequent project with the Province, or a successor Government of Ontario program; and
 - b. the license expires on the day that the Recipient does not hold a

valid agreement with the Province to carry out a project with the Province, or a successor Government of Ontario program; and to immediately cease using, modifying, reproducing or distributing the other products upon the expiry of the licence.

SCHEDULE "C"

PROJECT DESCRIPTION AND TIMELINES

The Project will be delivered for the period [ENTER the Effective Date of the agreement] to [ENTER end date of the agreement].

1. BACKGROUND

SkillsAdvance Ontario (SAO) is a sector-focused workforce development pilot that provides employers in specific sectors with access to job ready, skilled workers that meet their workforce development needs; and provides Individuals with sector-focused employment and training services, including experiential development opportunities to support them to obtain, succeed, and advance in employment.

SAO will support employers to:

- a. Build a supply of skilled Participants who can meet their workforce development needs by providing training services to improve their workforce performance;
- b. Reduce turnover and associated recruitment and training costs;
- c. Demonstrate commitment to workforce development, including advancement of Incumbent Workers; and
- d. Form effective and dependable workforce development partnerships with delivery partners.

SAO will provide Participants with:

- a. Training related to an occupation that offers positive labour market prospects;
- b. Essential, technical, and employability skills training, and on-the-job experience;
- c. Industry-recognized certificate or credential upon completion of training;
- d. New jobs in field of training;
- e. The ability to sustain or advance from their current employment; and
- f. Sustainable employment (i.e. sufficient hours to sustain household income; greater security through improved employment status and reduced chance of getting laid-off).

SAO sectors are identified by delivery partners and advisory committee (refer to Section 3 in this schedule for roles and responsibilities of delivery partners and advisory committee) in Ontario as having vacancies and advancement opportunities for partnership to develop sector-focused training curriculum and employment services to meet workforce development needs.

2. OBJECTIVES

SAO is intended to support:

- Sector-focused work-force development in all aspects of delivery, including skills training and pre-employment, employment, and post-employment services:
- b. Employers with hiring, onboarding and retaining employees and responding to employer skills requirements in the local economy; and
- c. Participants with succeeding and advancing in employment.

3. ROLES AND RESPONSIBILITIES

The Province will:

- a. Administer the Project by receiving and monitoring progress and performance against Agreement activities and outcomes;
- b. Collect and review products, outcomes, and data to evaluate the Project;
- Assess the Recipient's capacity to manage complex relationships, its knowledge of and credibility in the sector and regional community, as well as its project management capabilities;
- d. Conduct Recipient site visits; and
- e. Participate in advisory committee, as appropriate.

The Recipient will:

- a. Deliver SAO activities by undertaking all of the Project components and subcomponents;
- Lead delivery partners (such as service providers, training providers, SAO employers and community partners) and advisory committee to carry out Project components and subcomponents;
- c. Establish an advisory committee (if an advisory committee has not been established):
- d. Ensure advisory committee consists of representatives from employers, employment training providers, industry associations, labour and economic development organizations, and sector business development organizations:
- e. Ensure delivery partners and advisory committee:
 - Advise on sector requirements, including identifying job vacancies to be filled:
 - ii. Develop relevant sector-focused employment services and skills training curriculum;
 - iii. Maintain the ongoing relevance of the workforce development activities; and
 - iv. Deliver sector-focused employment services and skills training;
- f. Enter into and maintain contractual arrangements with delivery partners to deliver SAO;

- g. Develop and manage clear and transparent business arrangements and practices with delivery partners, including developing:
 - i. Clear performance expectations for delivery partners;
 - ii. Consistent and transparent methods for distributing operating Funds and financial supports to delivery partners;
 - iii. Practices for the sharing of Participant data between delivery partners and the Recipient. Participant data sharing practices must comply with applicable privacy legislation;
 - iv. Policies regarding service delivery and coordination, such as intake and assessment processes and procedures, referral processes;
 - v. Processes to collect and report data from all delivery partners in order to accurately complete reporting requirements as required;
 - vi. Written policies on the reimbursement payment process to all delivery partners;
- h. Participate actively in the reporting, monitoring and evaluation of the Project;
- i. Ensure that objectives, milestones and outcomes are being met;
- j. Keep a record of progress and accomplishments to-date and maintain accurate financial records;
- k. Ensure that funding is being used for its intended purpose and that objectives stated in the Agreement are being met;
- I. Provide prompt and accurate reimbursement to employers, according to the terms and conditions of the training or placement agreement; and
- m. Input and actively manage information in EOIS-CaMS as required under the Agreement.

4. EMPLOYER ELIGIBILITY AND SUITABILITY

The Recipient will ensure that employers meet the eligibility and suitability requirements set out below:

a. Eligibility:

The Recipient will ensure that employers:

- Have identified job vacancies or advancement opportunities in Ontario consistent with the workforce development needs to be addressed by the partnership;
- ii. Are registered and licensed to operate in Ontario;
- iii. Comply with all applicable legislation, including federal and provincial human rights legislation, regulations, and any other relevant standards, as well as the Occupational Health and Safety Act, R.S.O. 1990, Employment Standards Act, 2000, S.O. 2000, and Freedom of Information and Protection of Privacy Act, R.S.O., 1990;
- iv. Maintain appropriate Workplace Safety and Insurance Board or private workplace safety insurance coverage, as well as adequate

- third party general liability insurance as advised by its insurance broker;
- v. Provide job placements in Ontario;
- vi. Place the Participant on their payroll and provide the same employment terms, conditions, and benefits as for their regular employees during job placements;
- vii. Disclose any other government sources of funding associated with employing the Participants;
- viii. Do not hire Participants to displace existing staff or replace staff who are on lay-off;
- ix. Do not receive government funds from any other source for the same training and job placement services provided to the Participant; and
- x. Are not EO employment and training services providers.

b. Suitability:

The Recipient will ensure that employers:

- i. Demonstrate identified vacancies or advancement opportunities that have the potential to be permanent with long-term sustainability;
- ii. Offer support for training completion through:
 - 1. Flexible work arrangements and experiential development opportunities, such as job shadows or job placements;
 - 2. Providing adequate supervision and on-the-job training;
 - 3. Collaboration with delivery partners and advisory committee during the term of the Agreement in identifying specific essential, technical, and employability skills requirements associated with vacancies and advancement opportunities; and
 - Providing advice on changes required to sector-focused employment services, training curriculum or occupational training.

5. PARTICIPANT ELIGIBILITY AND SUITABILITY

a. Eligibility

The Recipient will ensure that Participants meet the eligibility requirements set out below:

- Individuals who are unemployed, precariously employed, or employed with low household income (for the purpose of this Project, Individuals who are working less than an average of 20 hours per week are considered to be unemployed and low household income is based on Low Income Cut-Off);
- ii. Individuals who are on a SAO employer's payroll but are facing barriers to job retention or advancement as identified by the delivery partners and advisory committee;

- iii. Individuals who are not participating in full-time training, education, or any other government training intervention that offers funding support for similar training or training-related costs (Note: Participants may access SAO financial supports in combination with supports from other government programs, provided they do not overlap or duplicate one another);
- iv. Residents of Ontario and legally entitled to work in Canada (i.e. citizens, permanent residents, or protected persons) and consistent with direction on other EO programs, Individuals who are not permanent residents or citizens (for example, 900-series Social Insurance Number holders) are not eligible, unless they meet the exceptions indicated in the Province's advisory on eligibility.
- v. 18 years of age or older; and
- vi. Individuals who are not employed in senior management or executive positions, or have controlling interest in the business.

b. Suitability

The Recipient will ensure that suitable Participants demonstrate the following factors:

- i. Having interest in pursuing a career in the identified sector; or for Individuals who are Incumbent Workers, having interest in advancement and demonstrate potential for success in SAO;
- ii. Poor employment history, for example, long-term unemployment, insufficient working hours (under-employment), or interruptions in work history;
- iii. Low household income, employment in a low-wage job, or dependence on non-employment income sources, such as social assistance or employment insurance benefits;
- iv. The ability to meet additional sector, workplace-specific requirements or occupational requirements as advised by the delivery partners and advisory committee, such as manual dexterity or strength; and
- v. The ability to meet additional partnership or community-specific criteria, for example, to support the employment and advancement of equity-seeking groups, as applicable.

6. PROJECT COMPONENTS

There are two Project components under SAO. The following outlines the two SAO project components and their subcomponents:

a. Partnership Development

- i. Partnership Building and Employer Needs Determination;
- ii. Partnership Sustainment (Management)

b. Employer and Worker Services:

- Sector-Focused Recruitment (Including Orientation and Service Planning)
- ii. Sector-Focused Pre-Employment Services ("Soft Skills Training")
- iii. Sector-Focused Essential and Technical Skills Training ("Hard Skills Training")
- iv. Sector-Focused Employment Services (Including Job Matching and Development)
- v. Sector-Focused Retention and Advancement Services and Ongoing Case Management

The Recipient will ensure that all of the activities under the following Project component(s) and its/their subcomponents are delivered:

Insert variable:

a. Partnership Development and b. Employer and Worker Services

or

a. Partnership Development

or

b. Employer and Worker Services

]

The Recipient may work with delivery partners to undertake any Project activities under the Project component(s) and subcomponents as required, or the Recipient may deliver all of the component(s) and subcomponents directly.

7. Partnership Development

- a. Partnership Building and Employer Needs Determination Under this project subcomponent, the Recipient will:
 - i. Conduct outreach activities with SAO employers to:
 - 1. Build trust across sector or industry employers and understanding of their alignment of interest; and
 - 2. Raise awareness of potential benefits of SAO and how it aligns with employers' business interests;
 - ii. Establish open and inclusive planning of workforce development priorities with sector and industry employers and associations, including the development of key success indicators by

- Complementing planning with effective governance, communication structures and approaches between SAO employers, advisory committee and delivery partners;
- iii. Develop and validate occupational and skills needs amongst SAO employers, which includes:
 - Mapping and inventorying in-demand occupations, required competencies and credentials and technical and essential skills requirements; and
 - 2. Identifying specific and projected vacancies or advancement opportunities across SAO employers;
- iv. Identify knowledge, experience and other resources, such as cash or in-kind contributions by SAO employers that can be leveraged to achieve shared goals and outcomes;
- v. Where applicable, enter into contractual arrangements with delivery partners and SAO employers for the development and delivery of employment and training services;
- vi. Develop employment and training service curriculum and delivery approaches that reflect knowledge of the sector's workforce development requirements through:
 - 1. Refocusing employment services to meet employability skills requirements (for example, sector-focused soft skills needs);
 - 2. Developing new training curriculum or revising existing curriculum to meet essential and technical skills needs; and
 - 3. Endorsing sector-focused employment and training services and curriculum by SAO employers and advisory committee;
- vii. Customize employment and training curriculum and delivery approaches to meet the needs of SAO employers by working with them to more fully articulate their training needs for the purposes of customization with the delivery partners and to ensure developed training and delivery approaches continue to meet SAO employers' needs on an ongoing basis; and
- viii. Conduct outreach with other community service providers to build awareness of SAO and to identify potential opportunities to develop connections that link Individuals to employment and training opportunities under SAO.

b. Partnership Sustainment (Management)

Under this project subcomponent, the Recipient will:

- Strengthen interpersonal and professional connections amongst SAO employers, advisory committee and delivery partners to encourage continued collaboration, coordination amongst stakeholders and to ensure Participants' needs are met;
- ii. Engage with delivery partners and advisory committee to modify and validate occupational needs and to update the competencies, credentials and skills required for occupations in the identified sectors;

- iii. Modify and refine employment service and skills training curriculum and delivery based on the changing needs of SAO employers; and
- iv. Build up sector and industry capability for partnering, which includes devising strategies, processes and systems for SAO employers to collaborate and partner.

8. Employer and Worker Services Component

Under this project component, the Recipient will ensure all of the services under all of the project subcomponents are made available to Participants and employers, but the Recipient is not required to deliver the subcomponents in the sequence listed below. Services may overlap with each other, based on needs and circumstance. The degree of support and intensity of each service will vary by Participant and not all Participants will need all subcomponents, or all of the activities within a subcomponent.

a. Sector-Focused Recruitment (Including Orientation and Service Planning):

The objective of this subcomponent is to identify Individuals with employers to receive training and services, to fill identified vacancies, and to lead them to retention in current employment or advancement opportunities.

The Recipient will:

- i. Recruit potential eligible and suitable Individuals for participation and provide orientation to the Individuals;
- ii. Create a case management profile;
- iii. Work together with delivery partners and SAO employers to screen Individuals based on:
 - 1. Suitability criteria developed for the SAO Project, such as specific physical requirements of the occupation;
 - 2. Identified workforce development needs such as aptitude for success in essential, technical, and employability skills training; and
 - 3. Potential fit with workplace culture as determined by the advisory committee; and
- iv. Screen Individuals for participation based on suitability (in particular, for Jobseekers, the Individual would not find employment in the identified sectors without SAO services).

Orientation Activities

The Recipient will:

 Conduct interview(s) to assess Individuals' motivation and attitude, employability and, where applicable, current job situation (for example, gauge interest in career advancement);

- ii. Ensure Individuals are made aware of SAO and sector-specific requirements at the employer's job site, if possible, such as:
 - The requirement to participate in service components based on their needs, including information on time commitments and sector's employment requirements; and
 - 2. Requirements and conditions of advancement opportunities in the sector, if already employed.
- iii. Assess essential skills and job readiness, based on SAO employer input and training requirements, and support evaluation of learning outcomes (pre- and post-training and intervention);
- iv. Notify Individuals that personal information will be used to contact them for the purpose of monitoring and follow-up; and
- v. Advise Individuals well in advance of additional invasive screening activities used for employment in certain sectors.

Case Management and Access to SAO

The Recipient will:

i. For the Individual who is eligible:

Ensure a case management profile, is created and completed in EOIS-CaMS for the Individual who wishes to proceed with the determination of suitability.

ii. For the Individual who is eligible but not suitable for SAO:

Ensure that the Individual is referred to or provided with information on employment and training programs and services.

iii. For the Individual who is eligible and suitable and wishes to proceed with SAO:

Ensure a service plan is created for Participants indicating registration in SAO.

Service Planning

The Recipient will ensure that:

 The service plan is developed jointly with the Participant and identifies goals as they relate to SAO participation, all Project components and subcomponents the Participant will take part in, any participation supports or referral services the Participant may require to achieve his or her goals, and is updated regularly, as the Participant moves through different Project components and subcomponents to achieve employment and career or advancement goals;

- 2. If the Participant is a social assistance recipient, consent to contact the social assistance case manager or administrator will be obtained in order to support service coordination as required;
- 3. If the Participant is a social assistance recipient, but does not provide consent to contact the social assistance case manager or administrator, advise the Participant that access to participant financial supports and job placement stipend under SAO will not be made available:
- 4. The Participant's case is managed throughout the Participant's involvement in employment and training services; and
- 5. Post-employment follow-up is conducted.

iv. Employability, Essential Skills and Job Readiness Assessment

Ensure that the Participant's employability, essential skills and job readiness are assessed using an assessment tool recommended by the advisory committee.

The Recipient will ensure that:

- 1. The assessment tool determines Participant's strengths and areas for improvement;
- 2. The assessment tool measures Participants' progress and their skill level pre- and post- training;
- 3. The same assessment tool is used for both the pre- and post-training assessments; and
- The Participants' pre- and post- assessment results of their employability and essential skills training are reported to the Province.

b. Sector-Focused Pre-Employment Services ("Soft Skills Training"):

The objective of this subcomponent is to deliver sector-focused pre-employment services and prepare workers for employment or advancement opportunities in the identified sectors.

The Recipient will ensure that Participants are provided with the following skills using curriculum developed or updated based on sector employers' needs as identified by the advisory committee:

- i. Sector-focused employability skills to obtain, retain and succeed in employment, such as:
 - 1. Career readiness or soft skills;
 - Job attainment skills, including preparing a résumé and cover letter, completing a job application, and succeeding in a job interview:
 - 3. Sector-focused job attainment skills that are identified by advisory committee;

- 4. Transferable employability skills required to support their longterm resilience in the labour market; and
- 5. Personal attitudinal and behavioural management skills for the workplace, such as teamwork and time management;
- ii. Sector-focused career planning and management skills to retain and advance in employment that can be obtained through activities such as:
 - 1. Individualized career counselling;
 - 2. Ongoing career planning;
 - 3. Personal attitudinal and behavioural management skills coaching such as positive attitude and responsible behaviours; and
 - 4. Sessions on worker rights, including employee duties and employer obligations under the *Employment Standards Act, 2000,* and the *Occupational Health and Safety Act, 1990*.

Curriculum Development Agreement(s) and Training Delivery Agreement(s)

Where the Recipient will develop or update a training curriculum, or deliver training by itself, the Recipient will keep the following documentation:

i. Curriculum Development or Revision

- 1. Name and description of the curriculum;
- 2. Date curriculum will be delivered to the Recipient;
- 3. Credential to be obtained (for example, a certificate);
- 4. Proof of SAO employer endorsement; and
- 5. Reports on Project and Participant progress

ii. Training Delivery

- Name and description of the training;
- 2. Duration of training:
- 3. Credential to be obtained (for example, a certificate);
- 4. Number of Participants to be trained;
- 5. Reports on Project and Participant progress; and
- 6. Proof of training delivery in Ontario.

Where the Recipient will work with delivery partners to develop or update the curriculum or deliver training, the Recipient will enter into curriculum or training delivery agreement(s) as outlined below, and provide payments as appropriate.

Curriculum Development Agreement(s):

If an existing curriculum requires revisions or if a new curriculum is to be developed, at a minimum, the curriculum development agreement(s) will include:

- i. Name and contact information of the curriculum developer;
- ii. Name and description of the curriculum;
- iii. Date curriculum will be delivered to the Recipient;

- iv. Credential to be obtained (for example, a certificate);
- v. Proof of SAO employer endorsement; and
- vi. A requirement for the curriculum developer to supply the Recipient with any reports on Project and Participant progress required by the Recipient to meet its reporting, monitoring and evaluation requirements in the Agreement.

Training Delivery Agreement(s):

At a minimum training delivery agreement(s) will include:

- i. Name and contact information of the training provider;
- ii. Name and description of the training;
- iii. Duration;
- iv. Credential to be obtained (for example, a certificate);
- v. Number of Participants to be trained;
- vi. Cost per Participant;
- vii. Training provider refund policies;
- viii. Requirement that the training provider give the Recipient any reports on Project and Participant progress required for the Recipient to meet its reporting requirements to the Province; and
- ix. Proof of training delivery in Ontario.

c. Sector-Focused Essential and Technical Skills Training ("Hard Skills Training):

The objective of this subcomponent is to provide Participants with sector-focused essential and technical skills training based on the needs of employers in the identified sectors (i.e. the skills required to obtain and succeed in the jobs for which vacancies, retention or advancement opportunities are identified by SAO employers and the advisory committee).

The Recipient will ensure Participants are provided with:

- Training that integrates essential and technical skills, as required to enter the jobs for which vacancies, retention or advancement opportunities are identified:
- ii. Any required workplace safety training (including certification) required;
- iii. Training opportunities that are short in duration (no longer than six months to accelerate entry to employment and generation of income); and
- iv. Training that will lead to an industry-recognized credential or certificate of completion endorsed by the delivery partners and advisory committee to support resilience in employment.

d. Sector-Focused Employment Services (Including Job Matching and Development):

The objective of this subcomponent is to provide Participants and employers with employment services including job search assistance, job matching and development, job shadowing, job placement, job coaching, screening support, onboarding and workforce retention support.

Participants

Under this subcomponent, the Recipient will ensure Participants are provided with:

- Job search assistance services, job coaching and support during the job search process;
- ii. Job matching and development services that identify Participants' skills and occupational interests and potential employment opportunities;
- iii. Job shadowing opportunities that will provide Participants with an understanding of how skills are applied in the workplace by observing existing employees at the workplace perform their jobs. This subcomponent will:
 - 1. Help Participants validate their work interests and skills and identify required workplace accommodation and training needs; and
 - 2. Be of a short-duration, as determined by the Participants and the SAO employer.
- iv. Job placement opportunities that will support Participants in bridging into sustainable employment through:
 - 1. Paid shorter-term experiential learning opportunities for Participants to practice and demonstrate skills learned in a workplace, replicating the work tasks associated with the job, as well as the opportunity to gauge whether the job is a good fit with their employment interests and goals;
 - 2. Offering SAO employers opportunities to gauge and evaluate Participants' skills, readiness for employment, and fit with the sector in advance of hiring; and
 - 3. Mentorship activities (for example, working with employers to identify and support mentors in the workplace, offering mentorship workshops and other activities as determined by the advisory committee).

Job placements are used in instances where there is potential to lead to permanent employment in an identified vacancy or advancement opportunity. Where job placements are used, the Recipient will work with the Participant and the SAO employer on establishing a placement agreement that details a plan and necessary supports required to achieve employment goals. Consistent with other Employment Ontario programs, exceptions may apply for Participants with disabilities.

The Recipient will ensure wages paid to the Participants during job placements meet all requirements under the *Employment Standards Act, 2000*.

Participants may participate in more than one job placement, depending on the vacancies identified by the delivery partners and advisory committee.

The Recipient will conduct at least one site visit prior to any job placements with each SAO employer to confirm the suitability and safety of the workplace.

Should a Participant accept an offer of employment prior to this subcomponent, the Participant is not required to participate in a job placement.

v. Job coaching that prepares Participants for job placements and employment before the placement; and after the placement is completed, coaching in the form of regular monitoring of progress and ongoing support.

The Recipient will ensure that employment goals stated in the placement agreements are met by both the Participant and the SAO employer and offer the following services as required:

- Guidance and direction on the employer's expectations and workplace standards;
- ii. One-on-one, on-site training and support, including orientation to workplace health and safety as required;
- iii. Identifying, preventing and resolving issues as they arise during employment;
- iv. Working with the Participant and the employer to identify and make use of internal and external training as well as career development opportunities that align with the Participant's employment and career goals; and
- v. Building the Participant's work capacity and support on-the-job activities through monitoring activities such as site visits.

Screening Activities

Once the Participant has been matched with job shadowing or job placement opportunities, the Recipient will coordinate the following screening activities in accordance with article 10 in Schedules A and B where the employer advises it is necessary and the Participant has provided consent to same:

- i. Testing for manual dexterity or strength;
- ii. Drug screening; and
- iii. Criminal background checks:
 - The employer reviews the criminal background/police records check to determine if it raises any relevant concerns about employment in the sector. If the record may present a barrier in the hiring process, but arguably does not raise any relevant concerns, the Recipient may advocate on behalf of the Participant (for example, demonstrating how they meet employers' identified needs);

- 2. If the record raises relevant concerns about employment in the sector, the Participant may not move forward with that particular job shadow, job trial, or job placement;
- 3. If the barrier is only with a particular employer, the Participant could potentially be matched to another employer in the sector;
- 4. If the barrier would apply across the identified employment sector, additional service planning by the Recipient and the Participant will be required to identify more appropriate employment goals.

Employers

Under this subcomponent, the Recipient will ensure employers are provided with sector-focused job matching and development services and other required employment services to support workforce development needs including:

- i. Screening services that support SAO employers in determining their workforce needs, screening and identifying Participants who are suitable matches based on their relevant skills, education and interests for any identified vacancies or advancement opportunities;
- ii. Coaching services where the Recipient acts as a liaison between the Participant and the employer, working with both sides to identify and resolve issues, as well as provides employers with assistance in evaluating and monitoring the Participant's job performance and progress; and
- iii. Onboarding and workforce retention support, such as recruitment support, onboarding new hires, and developing retention strategies that will help employers keep talented workers within their businesses.

e. Sector-Focused Retention and Advancement Services:

The objective of this subcomponent is to provide post-employment services to both Participants and employers for one year following the start of employment.

The Recipient will ensure that Participants are provided with sector-focused postemployment services that support Participants in retaining skills, succeeding, and advancing in employment in the identified sector. Post-employment services include ongoing individual case management, mitigation of risks to postplacement success and referrals to wraparound supports as required.

The Recipient will:

- i. Develop a follow-up plan with Participants upon conclusion of their placements, and identify a minimum number and frequency of communication to mitigate any potential risks to employment success;
- ii. Provide sector-focused career planning and management assistance, such as individualized career counselling and ongoing career planning; and
- iii. Sector-focused re-employment assistance, for example,

- 1. If a Participant is unable to retain employment with the employer, work with the Participant to identify more appropriate employment opportunities and address any barriers to retention; and
- 2. If a Participant is no longer interested in remaining in the sector, provide assistance to identify more appropriate employment opportunities outside of the sector, or refer the Participant to other EO Services.

8. FINANCIAL SUPPORTS

a. Employers

- i. The Recipient will ensure that financial supports:
 - Offset additional supervisory or training costs for mentors to provide supervision to Participants, such that employers will not need to make additional investments to train Participants hired through SAO;
 - 2. Help employers to ensure that conditions are in place to support placement success (for example, adequate integration and supervision or mentorship available to Participants);
 - 3. Are available to employers to support paid release time for Participants to maintain ongoing contact with the Recipient to identify and mitigate risks to their placement success;
 - 4. Do not subsidize the placement Participant's wages; and
 - 5. Are not available to public sector and broader public sector employers.
- ii. The maximum employer support amount per job placement is up to \$1,000 (details of employer supports are set out in Schedule "H" of the Agreement).
- iii. Are reflective of the duration and intensity of support based on need and in accordance with Schedule "H".

b. Participants

- i. The Recipient will ensure that financial supports:
 - Are available to Participants to return for employment services, for example, on a weekly or bi-weekly basis during job placements (details are set out in Schedule "H" of the Agreement);
 - 2. Are available to Participants to address temporary financial barriers to participation in SAO (details are set out in Schedule "H" of the Agreement);
 - 3. Are consistent with other EO programs, where exceptions may apply for Participants with disabilities as set out in Schedule "H" of the Agreement;
 - 4. In the form of job placement stipends (equivalent to the prevailing minimum wage for a maximum of 35 hours per week

- up to two weeks in duration) are available to Participants to support their first two weeks of a job placement, or employment, bridging the gap between their employment and their first pay cheque;
- 5. For each job placement is appropriate for each Participant, and reflective of the duration of support;
- 6. Do not exceed a maximum of \$3,000 for both Participant Financial Supports and the Job Placement Stipend; and
- 7. Are not provided to Participants who receive similar supports through other programs, for example under Ontario Works or Ontario Disability Support Program.
- ii. The Recipient will obtain Participant consent to contact and work with the social assistance case manager to ensure that Participants accessing social assistance programs:
 - 1. Will receive the maximum amount of support appropriate (for example, if the Participant is eligible to receive greater support through work-related benefits than through the Project); and
 - 2. Will not receive supports concurrently from other social assistance programs to address the same barrier.

Note: Incumbent Workers are not eligible for the job placement stipend.

16. EOIS-CaMS

The Recipient will ensure that EOIS-CaMS is used to support the delivery of SAO, case management of Participants, and reporting of Participant and employer information to the Province.

The Recipient will:

- a. Comply with the Terms and Conditions of the SPRA EOIS-CaMS Registration and any directions and policies provided by the Province relating to the use of EOIS-CaMS:
- Use EOIS-CaMS in accordance with the EOIS Case Management System User Management, and EOIS-CaMS Service Provider User guides posted on the EOPG;
- Be responsible for all staff who are authorized to use EOIS-CaMS and ensure that only authorized staff have access to EOIS-CaMS and the Reporting Website;
- d. Use EOIS-CaMS to manage Participant cases and report Participant and employer data/information in a timely manner;
- e. Designate at least two staff in their organization who will be assigned the role of service provider registration authority (SPRA);
- f. Ensure that the SPRA is responsible for setting-up and maintaining

access for authorized staff.

17. CUSTOMER SATISFACTION

The Recipient will:

- a. Provide 100% of Participants and employers accessing SAO services with a customer satisfaction survey to complete at exit; and
- b. Conduct the survey using questions provided by the Province.

18. MONITORING AND EVALUATION

a. Monitoring

Monitoring, under the Agreement, will be a collaborative effort between the Province and the Recipient. Its aim is to support continuous improvement. Some of the activities that can be expected during the life cycle of an Agreement include:

- i. Activity and financial reporting;
- ii. On-site compliance and evaluation visits;
- iii. Discussions via telephone; and
- iv. Correspondence by mail or e-mail.

b. Evaluation

If requested by the Province, the Recipient will assist the Province in its evaluation and performance management of the Project by:

- i. Contacting Participants and employers on behalf of the Province;
- ii. Contacting other stakeholders on behalf of the Province; and
- iii. Participating in evaluation activities conducted by or on behalf of the Province including surveys, interviews and discussion groups.

19. CUSTOMER SERVICE

Without limiting the generality of section 2.3 of the Agreement, the Recipient shall have comprehensive program management systems that include policies and procedures to ensure the delivery of quality customer services as well as adequate and appropriate human resource, management and administrative support.

The Recipient shall:

- a. Have:
 - i. a customer service charter that is posted and visible to customers;
 - ii. delivery site(s) and facilities and days/hours of operation that are posted and visible to customers;
 - iii. a customer complaint and resolution process in place.

- b. Have, at a minimum, the following three elements in their customer service charter:
 - i. believe in quality service;
 - ii. encourage feedback (compliments or complaints);
 - iii. will follow up on this feedback in a prescribed manner and timeframe.
- c. Ensure that delivery site(s) and facilities reflect customer need, including:
 - i. accessible facilities or service provision at an accessible site;
 - ii. itinerant and/or mobile services where local need is identified:
 - iii. days or hours of operation that reflect customer need.
- d. Obtain the prior written consent of the Province to adjust their delivery site(s), facilities, or days and hours of operation.
- e. Ensure that extended evening and weekend hours of service are offered where there is identified need.
- f. Not be required to offer these services on statutory holidays.

20. DOCUMENTATION REQUIREMENTS

a. Employers:

The Recipient will ensure that its records relating to an employer contain:

- i. Proof of employer's eligibility;
- ii. Documentation of at least one site visit per employer, prior to any placements, to confirm that the employer can provide a suitable and safe workplace environment:
- iii. Employer declaration of WSIB or alternative workplace safety insurance coverage and third party liability insurance;
- iv. A completed SkillsAdvance Ontario Employer Registration form;
- v. Employer profile information, including size, sector and the Participant's occupational profile;
- vi. Information that describes the capacity of the employer and the workplace to provide relevant training;
- vii. A completed, signed copy of the SkillsAdvance Ontario Training Placement agreement;
- viii. An assessment of the employer's ability to fulfill placement commitments and to provide employment to the Participant after the training is completed which may include:
 - 1. Evidence of employer's request for placement supports;
 - 2. Information on placement duration(s); and
 - 3. Evidence of Recipient approval and payment of supports based on the job placement.

b. Participants:

The Recipient will ensure that records for Participants include:

i. A completed, signed and dated SkillsAdvance Ontario Participant Registration form which contains the Province's notice of collection

and the consent to the Province's indirect collection of personal information:

- ii. A statement on file that
 - 1. The Participant's photo identification has been reviewed; or
 - 2. Two pieces of government-issued identification have been reviewed (for example, social insurance number card, birth certificate), if the Participant does not have photo identification available; and
 - 3. Proof of eligibility to work in Ontario has been reviewed (for example, permanent resident card, social insurance card);
- iii. Assessments, service referrals and evidence of progress;
- iv. A clear and achievable service plan for employment and or training;
- v. Rationale to support SAO participation, including suitability and employability information;
- vi. Clear explanations of how service is expected to improve employability;
- vii. Details of financial supports, such as:
 - 1. Reason for the supports;
 - 2. Amount of supports;
 - 3. Confirmation of income level; and
 - 4. Authorized signature for supports.

21. FACILITIES AND FACILITIES LEASES

a. Facilities

The Recipient will:

- i. Have the ability to deliver the Project in a facility that is readily accessible to all persons, including persons with disabilities; and
- ii. Where services are not fully accessible to persons with a physical disability, have a plan to accommodate these persons by serving them in an accessible location and/or through partnership with another organization.

b. Facility Leases

If relocation or revision of facility arrangements is required, the Recipient will obtain prior written approval from the Province to use the Funds for this purpose prior to entering into agreements or making financial commitments.

22. FRENCH LANGUAGE SERVICES

The Recipient will ensure that the Project is carried out in French, as directed by the Province, including providing the following in French:

- a. Outreach
 - Marketing materials (brochures);
 - ii. Outreach strategies developed and conducted;
- b. Verbal Communications
 - i. Telephone;
 - ii. In person, such as interviews, visits, meetings, workshops and information sessions and consultations;
- c. Written Communications
 - i. Correspondence such as letters and faxes;
 - ii. Email, interactive databases and Internet;
- d. Signage and Public Notices
 - i. Interior and Exterior;
- e. Forms and Documents
 - i. Stationery;
 - ii. All forms used for identification, certification or application such as licenses, and certificates; and
 - iii. Any document intended for public use.

If the Recipient is not directed by the Province to carry out the Project in French, the Recipient will ensure that individuals requiring French language services are referred to a designated service provider.

23. FORMS

The Recipient will use forms identified by the Province including the following:

- a. SkillsAdvance Ontario Participant Registration form:
- b. SkillsAdvance Ontario Employer Registration form; and
- c. SkillsAdvance Ontario Training Placement Agreement.

These forms are mandatory and cannot be altered by the Recipient. These forms are available on the EOPG.

The Recipient will develop a separate document if it requires additional information or consents from the Participant.

24. INSURANCE

The Recipient will ensure that:

- a. Employers have sufficient insurance coverage purchased to meet the requirements. Details of the requirements are set out in the Employment Service Guidelines posted on the EOPG under sections 5.11 and 5.12.
- b. Employers comply with the insurance requirements and claims procedures and processes; and
- c. Participants and employers are aware of their workplace safety responsibilities.

25. TRANSFER PAYMENT COMMON REGISTRATION SYSTEM

The Transfer Payment Common Registration (TPCR) system is a mandatory central repository for organizations to register, view, and update their information in order to receive transfer payment funding from the Province. In order to establish a profile in the TPCR, the Recipient will follow the instructions posted on the TPCR system website to:

- a. identify an individual to function in the Administrator role;
- b. identify any additional individuals (Associate Users) who should have access to the organization's profile;
- c. verify the pre-populated information within the TPCR system;
- d. make any additions or edits required; and
- e. keep information up-to-date.

SCHEDULE "D"

BUDGET

Sector [ENTER Sector Name]

2019-2020

FUNDING CATEGORIES	AMOUNT
Operating Funds	\$
Financial Supports for Employers and Participants	\$
MAXIMUM SITE BUDGET	\$

SCHEDULE "E"

PAYMENT PLAN

The monthly payment amount will be calculated as follows:

Maximum Funds / number of monthly payments *

*The number of monthly payments will be the number of calendar months for which there is Project activity for fiscal year(s) as specified in Schedule "C" Project Description and Timelines.

The last month's payment under this Schedule is subject to a hold back of 15% of the Maximum Funds. If the Funds paid under the Agreement are less than the Maximum Funds, the scheduled payment amount (and hold back amounts) will differ. The hold back will remain at 15% of actual Funds paid.

The funds subject to this hold back will be released by the Province, subject to adjustments as contemplated by the Agreement, upon Project completion. The Project is considered complete when the Province has received and reviewed all reports outlined in Schedule "F"

Pursuant to section 4.2c of the Agreement, the Province may adjust the monthly payment based on the Estimate of Expenditure Reports referenced in Schedule "H". For greater certainty, the Maximum Funds shall not be exceeded.

SCHEDULE "F"

REPORTING

ACTIVITY REPORTS

1.

For existing projects continuing into 2019-20, use:

Quarterly Status and Adjustment Report (QSAR)

Report 1 April 1, 2019 to June 30, 2019

Due Date: July 19, 2019

Report 2 July 1, 2019 to September 30, 2019

Due Date: October 21, 2019

Report 3 October 1, 2019 to December 31, 2019

Due Date: January 17, 2020

Report 4 January 1, 2020 to March 31, 2020

Due Date: April 20, 2020

For new projects/new recipients starting in 2019-20, use:

SkillsAdvance Ontario Pilot Start-up Progress Report and Compliance Questionnaire

For Effective Date to June 30, 2019

Due Date: July 19, 2019

Quarterly Status and Adjustment Report (QSAR)

Report 1 July 1, 2019 to September 30, 2019

Due Date: October 21, 2019

Report 2 October 1, 2019 to December 31, 2019

Due Date: January 17, 2020

Report 3 January 1, 2020 to March 31, 2020

Due Date: April 20, 2020

FINANCIAL REPORTS

2. Estimate of Expenditure Report (EER)

Report 1 for Effective Date to July 31, 2019

Due Date: August 13, 2019

Report 2 for Effective Date to September 30, 2019

Due Date: October 10, 2019

Report 3 for Effective Date to November 30, 2019

Due Date: December 11, 2019

Report 4 for Effective Date to December 31, 2019

Due Date: January 13, 2020

Report 5 for Effective Date to March 31, 2020

Due Date: April 3, 2020

3. Statement of Revenue and Expenditure Report (SRER)

For the period Effective Date to March 31, 2020

Due Date: June 12, 2020

4. Auditor's Report, if applicable

For the period Effective Date to March 31, 2020

Due Date: June 12, 2020

5. Reports specified from time to time

On a date or dates specified by the Province.

6. Report Details

All reports will be submitted on the dates set out above. All reports above, except for the Auditor's Report, will be completed using a template provided by the Province. The templates will be available on the EOPG.

Requirements for the Auditor's Report are set out in Schedule "H".

SCHEDULE "G"

PERFORMANCE COMMITMENTS

Sector [ENTER Sector Name *NOTE: Repeat the entire sections of "INTAKE, CUSTOMER SERVICE, and DELIVERABLES in this Schedule should there be more than one sector]

2019-2020

*INTAKE

Participant (Jobseekers) Intake Target	
Participant (Jobseekers) Placement Target	
Participant (Jobseekers) Retention Target	
Participant (Incumbent Workers) Intake Target	
Participant (Incumbent Workers) Placement Target	
Participant (Incumbent Workers) Retention Target	

^{*}Intake Targets are defined in Schedule "B"

CUSTOMER SERVICE

PERFORMANCE COMMITMENTS	MINIMUM STANDARD	
Customer Satisfaction (Participant/Employer)	85%	

PARTNERSHIP DEVELOPMENT DELIVERABLES

ACTIVITIES	COMPLETION DATE
Establish SAO Advisory Committee Governance structure	
Develop a Sector-focused Recruitment plan	
(including orientation and service planning)	
Develop sector-focused curriculum	
Develop a sector-focused pre-employment	
services ("soft skills training") framework	
Develop a sector-focused essential and	
technical skills training ("hard skills training")	
framework	
Develop job matching and job placement	
plan/arrangements	
Develop sector-focused employment services	
and post-employment services plan	

EMPLOYER AND WORKER SERVICES DELIVERABLES

ACTIVITIES	COMPLETION DATE
Lead delivery partners and SAO Advisory	
Committee to deliver Project activities	
Deliver sector-focused recruitment activities	
(including orientation and service planning)	
Deliver sector-focused pre-employment services	
("soft skills training")	
Deliver sector-focused essential and technical	
skills training ("hard skills training")	
Deliver job matching and job placement activities	

SCHEDULE "H"

AUDIT AND ACCOUNTABILITY REQUIREMENTS

Audit and Accountability Requirements For Effective Date to March 31, 2020

Note to Auditors and other readers: Schedule "H", the Audit and Accountability Requirements, must be read in conjunction with the Agreement and its other Schedules.

1. Definitions

In this Schedule, the following terms have the following meanings:

- a. "Capital Assets" means capital assets (tangible and intangible), are identifiable assets that meet all of the following criteria:
 - are held for use in the provision of services, for administrative purposes, for production of goods or for the maintenance, repair, development or construction of other capital assets;
 - ii. have been acquired, constructed or developed with the intention of being used on a continuing basis;
 - iii. are not intended for sale in the ordinary course of operations; and
 - iv. are not held as part of a collection.

Tangible capital assets include land, buildings and equipment. Intangible capital assets are identifiable non-monetary assets without physical substance.

- b. "Multi-agreement holder" means a Recipient that has two or more agreements with the Province within a single region or across more than one region.
- c. "Project Accounting" means that, where a Recipient has multiple projects, all costs will be allocated to a specific Project. These costs will then be reconciled with the sources of funding, ensuring accurate accounting. The Recipient may receive funding from multiple sources. For the purpose of Audit and Accountability Requirements, each Project has a separate source of funding. Project Accounting connects Recipient funding with Project activities to effectively track the financial progress of a Project.

2. Funding Categories

a. Operating Funds

Operating Funds are Funds used for direct expenditures related to the day-to-day direct delivery of all of the components of the Project as agreed to with the Province.

No more than 20% of the Operating Funds can be used for the Partnership Development and Sustainment (Management) component.

- Costs related to the provision of the Project that can be claimed against Operating Funds include:
 - 1. Staff and management salaries directly associated with the delivery of the Project;
 - 2. Hiring and training of staff (including professional development);
 - 3. Marketing (signage, print/paper/web ads, outreach, etc.);
 - 4. Facilities (rent);
 - 5. Facilities (mortgage payments, but only the interest portion of a mortgage payment is allowed as an operating cost);
 - 6. Funding of Projects undertaken by the Recipient, including the distribution of funds to relevant partners;
 - 7. Accounting services and auditor's fees
 - 8. Administrative costs (as described below, up to a maximum of 15% of the Operating Funds)
 - 9. Partnership Development and Sustainment (Management) (up to a maximum of 20% of Operating Funds)
 - 10. Sector-focused recruitment (including orientation and service planning)
 - 11. Pre-employment service delivery ("soft skills training")
 - 12. Essential and technical skills training ("hard skills training")
 - 13. Employment services (including job matching and placement)
 - 14. Retention services and ongoing case management (post-employment services): and
 - 15. Other direct operating expenditures related to delivery of the Project.
- ii. The Recipient will not use Operating Funds for:
 - 1. Termination, severance costs and bonuses; or
 - 2. Major capital expenditures, such as the purchase or construction of facilities.

b. Administration Costs

Recipients are able to attribute no more than 15% of Operating Funds for administration costs. Administration costs are indirect expenditures necessary for operating an organization but not directly associated with the delivery of the Project. For example, a portion of the salaries/benefits of the Executive Director, IT, and/or financial staff that work for the entire organization but may spend a portion of their time dedicated to administrative functions that support the Project may be claimed as Administration Costs.

- c. Financial Supports for Employers and Participants are incremental supports to encourage employer and Participant participation. They are intended to support employers to ensure that the right workplace conditions are in place to support placement Participant success (i.e. adequate integration and supervision/mentoring of the placement Participant); and to support job placement Participants to maintain ongoing contact with service providers to identify and mitigate risks to their placement success.
 - i. **Employer Supports** are funds (up to \$1,000 per placement) used to offset the cost of:
 - ii. **Lost productivity of workplace mentors** assigned to assist Participants participating in a formal job placement.
 - 1. The rate for financial support related to mentorship is the assigned mentor's hourly or pro-rated wage, to a maximum of \$20/hour.
 - 2. The maximum amount payable to support workplace mentorship is equal to 26 hours x the maximum hourly wage of the assigned mentor, the equivalent of up to two hours per week for the three month recommended minimum duration of the job placement.
 - iii. **Paid release time** for Participants participating in a formal job placement to return to service providers for employment services on a regular basis (e.g. weekly or biweekly).
 - 1. The rate for financial support related to paid release time is equal to the Participant's hourly or pro-rated wage, to a maximum of \$20/hour.
 - 2. The maximum amount payable to support paid release time is equal to the Participant's assigned hours of participation in employment services during regular business hours over the course of a job placement, up to 48 hours.
 - iv. **Participant Supports** are funds (up to \$3,000 per Participant) available to address Participants' temporary financial barriers to participation in service components (from intake to completion of job placement).

- Participant Financial Supports: These are available to all Participants (both Jobseekers and Incumbent Workers) under the following categories:
 - a. Transportation;
 - b. Food:
 - c. Work clothing or uniforms;
 - d. Special equipment for work, such as workplace accommodation needs;
 - e. Emergency or occasional dependent care;
 - f. Employer specific assessments, such as academic assessment, certification charges, language assessment; and
 - g. Translation of academic documents.

Consideration of financial supports for Incumbent Workers is based on their current wage rates and the financial burden that is being incurred over and above their regular employment (for example, financial supports cannot be for costs that the Participants already bear with their regular employment)

2. Job Placement Stipend: These are available only to Participants who are Jobseekers to support the first two weeks of job placements or employment to bridge the gap between the start of their employment and their first pay cheque. Job placement stipend will be provided by the Recipient to the Participants as financial support, if required. Job placement stipend is not intended to subsidize or supplement wages for the job placements paid by the employers and is equivalent to the prevailing minimum wage for a maximum of 35 hours per week, up to two weeks in duration.

3. Financial Considerations a. Funds

- i. Recipients' use of Funds is also subject to the following limits:
 - 1. Operating Funds are allocated against an identified level of service;
 - In situations of co-location of the Project with other programs and services, Project funds must only be used to cover costs directly related to the delivery of the Project; this must be managed by applying Project Accounting principles;
 - 3. The Recipient will not transfer funds between budget lines (as set out in Schedule "D" unless it obtains the prior written consent of the Province; and
 - 4. The Recipient **should not anticipate** additional funds, although the Recipient should discuss any issues with the Province;
 - 5. Leasehold improvements directly related to the delivery of the Project require prior written approval from the Province.

b. Interest Earned

The Agreement sets out the requirements on Interest Earned.

- i. If the Recipient fails to identify interest earned in the Estimate of Expenditure Report EER(s) and the Statement of Revenue and Expenditure Report, then the following applies:
 - The Province will deem interest to have been earned based on the average
 of unspent Funds reported on the EER(s) and the SRER using the current
 interest rate charged by the Province of Ontario on accounts receivable.
 This deemed interest amount will be considered revenue of the Recipient
 for the purposes of the EER(s) and the SRER; and
 - 2. Such failure will be considered an Event of Default in accordance with section 15.1 of the Agreement.

c. Bank Account

The Agreement sets out the requirements regarding deposit of the Funds into an interest bearing account in section 4.6 of the Agreement.

Although maintaining a separate bank account for Funds received from the Province is not a requirement under the Agreement, it is a strongly recommended practice.

d. Capital Assets

The Recipient is not required to report capital asset expenditures to the Province. However, the Recipient will maintain a separate record of capital asset expenditures for audit purposes in accordance with Article 7.2 of the Agreement.

e. Disposition of Assets

The Agreement sets out the requirements on Disposition of Assets.

Any money earned on the disposition of assets will be reported on the SRER (and any other reports specified by the Province) in the period when disposition of the asset occurs.

f. Deficits

The Recipient is accountable for managing the Funds and required to remain within its approved site allocations (per Schedule "D" of the Agreement). The Recipient will request prior written approval from the Province for an anticipated over-expenditure. Each request will be individually evaluated. The Recipient will

enclose copies of overspending approval documents from the Province with their financial reports (for example, EER or SRER as requested by the Province.

g. Tax Rebates

Reporting Province program expenditures net of tax rebates:

The Recipient will report all program expenditures net of any tax rebates or input tax credits.

h. Sample

Amount Recipient spent on goods/services		\$100.00
Amount of tax paid (example 13%)	\$13.00	
Less amount of tax rebate claimed (where rebate equals 80%)	<u>-\$10.40</u>	
Amount of tax expenditure	\$ 2.60	
Amount reported as Province Project		<u>\$102.60</u>

4. Instructions for Reports Required

expenditure

Reports are complete if they are signed by a person with authority to bind the Recipient. Payments will be delayed if complete Reports are not received by identified due dates as directed by and in the form requested by the Province posted on the Employment Ontario website.

a. Estimate of Expenditure Report (EER)

The Recipient will submit to the Province Estimate of Expenditure Reports (EER) as required in accordance with the instructions set out in Schedule "F". Each EER must identify the total expenditure for the defined period for each site using the best information available at the time, as well as the forecast of expenditures at March 31 of each year of the Agreement.

The Province will provide the Recipient with a template for the EER posted on the EOPG. The Recipient will complete the EER as directed by and in the form requested by the Province.

The Province may adjust payments based on each EER received during the year.

b. Statement of Revenue and Expenditure Report (SRER)

The Recipient will submit to the Province Statement of Revenue and Expenditure Reports (SRER) as required **and in accordance with the instructions set out in Schedule** "**F**". The SRER is used as the basis for the Province's annual Reconciliation Report.

The Province will provide the Recipient with a template for the SRER posted on the EOPG. The Recipient will complete the SRER as directed by and in the form requested by the Province.

The SRER will verify that:

- i. Funds have been applied to costs directly related to the Project;
- ii. Funds received or expended in prior years for a similar project have not been included;
- iii. Funding and expenditures from other sources have not been included in the Report;
- iv. Project reported expenditures are net of tax rebates, credits and refunds referred to in section 4.9 of the Agreement;
- v. Shared costs have been properly apportioned to the Project using Project Accounting principles;
- vi. Interest earned on Funds has been credited to the Project;
- vii. Money earned on the disposition of assets has been credited to the Project and maintained in an interest-bearing bank account; and
- viii. Funds that were provided to the Recipient before the Recipient's immediate need for them were placed in an interest-bearing bank account in the name of the Recipient at a Canadian financial institution.

c. Auditor's Report

The Recipient will obtain an Auditor's Report when the Maximum Funds (set out in Schedule "B" of the Agreement) total \$150,000 or more. The audit of all SRERs will be conducted by an external auditor in accordance with Canadian Generally Accepted Auditing Standards. The auditor should verify, at a minimum, that proper and distinct accounts and records are maintained for program funds/expenditures by each budget line. The Auditor's Report will include an opinion on the SRER.

A copy of the full Auditor's Report will be submitted to the Province directly from the auditor.

An Auditor's Report is not required for Recipients who are:

- i. District School Boards or school authorities as defined in the Education Act, R.S.O. 1990, c. E2, as amended;
- ii. Universities established in Ontario that receive regular and ongoing operating funds from the government for purposes of post-

- secondary education;
- iii. Colleges of Applied Arts and Technology as defined in the Ontario College of Applied Arts and Technology Act, 2002. S.O. 2002, c. 8 Sch. F, as amended; or
- iv. Municipalities in Ontario as defined by the Municipal Act, 2001. S.O. 2001, c. 25, as amended.

Important Note: If additional details or corrections are necessary on the SRER, the Recipient's external auditor must verify any revisions in writing. The Province cannot accept unaudited information provided by a Recipient in support of their SRER.